doi: 10.1093/jeg/lbab019

The Geography of Knowledge and R&D-led Growth

Marta Aloi*, Joanna Poyago-Theotoky^{†‡}, Frédéric Tournemaine[§]

Abstract

We analyse how spatial disparities in innovation activities, coupled with migration costs, affect economic geography, market structure, growth and regional inequality. We provide conditions for existence and uniqueness of a spatial equilibrium, and for the endogenous emergence of industry clusters. Spatial variations in knowledge spillovers lead to spatial concentration of more innovative firms. Migration costs, however, limit the concentration of economic activities in the most productive region. Narrowing the gap in knowledge spillovers across regions raises growth, and reduces regional inequality by making firms more sensitive to wage differentials. The associated change in the industry concentration has positive welfare effects.

Keywords: Growth, Economic geography, Geographic labour mobility, Innovation, R&D,

Knowledge spillovers, Regional economics **JEL classifications:** O41, O31, L13, J61, R32

Date submitted: 30 September 2019 Editorial decision: 12 April 2021 Date Accepted: 23 April 2021

1. Introduction

Firm market power has increased across all sectors in the US, as recently documented by De Loecker and Eeckhout (2018), with the rise in mark-ups mainly driven by few firms concentrated in the upper tail, and market shares moving from low to high mark-up firms (except for the retail industry). This trend is most evident in dynamic industries where innovation is increasing. In terms of location, innovative activities in most countries are concentrated in few productive regions. One of the primary reasons why innovation tends to cluster spatially is that knowledge spillovers are stronger in specific places. The link between innovation and the spatial dimension of knowledge spillovers has received significant attention in economics and economic geography (see Feldman and Kogler (2010)). Less attention has been paid to the role played by localised knowledge externalities on the transmission of innovation on market structure, the dynamics of competition and growth.

The research question we consider is: How are regional disparities² in innovation activities shaping the geography of the economy, its market structure, the growth rate and inter-regional

[†]The Australian National University, Canberra, ACT, Australia; e-mail: joanna.poyago-theotoky@anu.edu.au

[‡]University of East Anglia, Norwich, England, UK; e-mail: *j.poyago-theotoky@uea.ac.uk*

[§]Chulalongkorn University, Bangkok, Thailand; e-mail: frederic.t@chula.ac.th

^{*}Corresponding author: School of Economics, University of Nottingham, England, UK; e-mail: marta.aloi@nottingham.ac.uk

¹ Most R&D intensive activities, such as ICT, pharmaceuticals and biotechnology, and automobiles are concentrated in few companies. For instance, the 2018 EU Industrial R&D Investment Scoreboard shows that in countries like Germany, France, Japan and the US the top 100 companies account for half of R&D spending.

² Our use of the term "regional/spatial disparities" differs from the traditional way it is used in the field of economic geography. We refer to spatial disparities as relating to innovation activities and differences in technology. In contrast, in the majority of the economic geography field "spatial disparities" refers to disparities in income differences and inequality refers to (income) differences among households.

inequality? We investigate this question within a setup which encompasses industry location and R&D-led growth with three distinctive features: (i) an endogenous market structure characterised by oligopolistic firms conducting R&D, (ii) spatially constrained knowledge flows and (iii) migration costs. We find that a reduced knowledge spillover deficit across regions leads to higher growth, reduced inter-regional inequality, and overall higher welfare.

We depart from the existing literature in three ways. Firstly, innovation clusters and regional disparities emerge endogenously, hence relating industry concentration and growth. Secondly, we can explain dispersed innovation activity without relying on transport (or congestion) costs, hence emphasising frictions in knowledge flows as the source of agglomeration economies. Thirdly, dispersion is robust to the possibility of migration.

The spatial disparities in technology (knowledge) spillovers encourage the spatial concentration of industries in the most productive region, however, inter-regional migration costs limit the geographical concentration of economic activities. As a result, not all firms necessarily operate in the most productive region. The oligopolistic (Cournot) market structure generates a positive relationship between concentration and market power, with the implication that the spatial concentration of industries is associated with fewer, but more R&D intensive, firms. Individuals endogenously choose their location, implying that the share of population in each region is affected by asymmetries in knowledge externalities across regions.

We find that higher knowledge spillovers increase Lao's demand in R&D, which in turn increases R&D costs (and, thereby, the entry level of R&D), leading to a reduction in the number of operating firms, and a higher mark-up. In equilibrium, the economy has fewer but more productive firms; it has a higher growth rate, and this effect is more pronounced for the lagging region. Intuitively, since labour costs are higher in the advanced region, more firms set up operation in the lagging region when knowledge spillovers increase, than in the advanced region for an equivalent increase in knowledge spillovers there. In this case, regional disparities in innovation clusters (and income) can also be reduced.

Welfare in each region depends on static components related to market imperfection and the relative wage, and on dynamic components associated with the rate of growth. We show that the growth gains associated with fewer and more productive firms outweigh the static losses associated with a lower degree of competitiveness (higher prices). Furthermore, we show that, under mild parameter restrictions, both regions experience welfare gains as spillovers in the lagging region strengthen. In essence, reducing regional disparities and promoting growth can be mutually compatible.

We provide novel insights on the role played by localised knowledge externalities on the emergence of regional disparities in innovation clusters, industry concentration and growth. The ability of advanced regions to better capture knowledge and ideas translates into more labour allocated to R&D, higher wages, higher productivity and higher spatial concentration of industry. The latter, in turn, influences the industry market power and the associated R&D and innovation activities. This prediction is consistent with empirical evidence, and deserves further exploration.³ Compared to monopolistic competition models

³ Recent work by Rossi-Hansberg et al. (2018) studies whether the positive trend in product market concentration, observed at the national level, persists at the local market level. They document heterogeneous trends across industries, with diverging trends prevailing in the retail industry, while in the manufacturing industry they observe mainly rising concentration both at the national and local level. The key mechanism behind the divergent trends in some industries rests on large enterprises opening up more plants in new local markets, thereby expanding the local number of establishments in the industry. This occurs particularly in sectors where transport costs are relevant (e.g. retail). Their results, therefore, do not contradict ours, which focus on R&D intensive (manufacturing) activities and negligible transportation costs.

with endogenous mark ups due to variable demand elasticities, we focus on models that generate higher market concentration and a rise in mark ups.

Our results also have potentially interesting implications for policy. Particularly, which type of intervention facilitates knowledge spillovers on local economies and a realignment in the spatial disparities in innovation activities across regions. One natural candidate would be to implement place-based university/educational policies. Empirical evidence indicates a casual link between university research activity and productivity in neighbouring firms, as exposure to research and inventors facilitate innovation (Jaffe (1989); Kantor and Whalley (2014)). Going a step further, lagging regions may also need new policies aimed at increasing exposure to innovation during childhood, as new evidence shows that more exposure to inventors and innovation in early life is key to increased innovation Bell et al. (2019).

Throughout the paper we concentrate on spillovers across regions rather than technological gaps (or absorptive capacity). Notwithstanding, our work naturally extends to economies not lying on the technological frontier. ⁵Finally, we discuss how our model may be extended to allow for population growth and firm heterogeneity. Specifically, firms belonging to a given sector are identical, while firm productivity differs across sectors. As a result, average knowledge between regions differs, mitigating the scale of firm relocation and inter-regional income and utility changes associated with a reduction in the gap in knowledge spillovers between regions.

Relation to the literature

The literature, pioneered by Krugman (1991), and further developed by the new economic geography discipline (Fujita et al. (1999); Martin and Ottaviano (1999); Ottaviano and Robert-Nicoud (2005)), has traditionally focused on spatial frictions to account for the spatial distribution of industries and the geography of economic activity. Yet, localisation persists despite the rapid decline in the costs of shipping goods and in communication costs (Glaeser and Kohlhase (2004); Head and Mayer (2004)). Alternative explanations, emphasising knowledge diffusion and learning, may be as relevant in accounting for clustering of innovation activities (Audretsch and Feldman (2004)). Regional variations in localised knowledge spillovers have been identified in the empirical literature since the seminal work of Jaffe et al. (1993), lately revitalised by Thompson and Fox-Kean (2005). Notably, it is the non-codified (tacit) type of knowledge that flows more easily locally than over great distances (Gertler (2003)). This is the type of knowledge that is transferred through person-to-person interaction and is clearly facilitated by geographic proximity and hindered by the costs of people moving (Combes and Duranton (2006)). Recent work by Bloom et al. (2013), estimate that the social return to R&D is two to three times higher than the private return. Building on Bloom et al. (2013); Lychagin et al. (2016) find that both intra- and inter-regional spillovers matter. Furthermore, they find empirical support for the hypothesis that reduced face-to-face knowledge flows account for the weakening of cross-regional spillovers in space. This evidence also fits with a vast body of literature

⁴ Focusing on the geography of innovation, Bell et al. (2019) show that direct exposure to a culture of invention and to role models appears to be playing a key role. Crucially, they show that the geographical aspects hold regardless of where you live as an adult, indicating the importance of early exposure. The authors point to the need to expose young people, especially those who show early-life excellence at math and science, to actual inventors and their workplaces, by promoting schemes ranging from mentoring by current inventors to internship programs at local companies

⁵ In appendix 9.8 we show that incorporating both spillovers and technological gaps would not alter our qualitative results.

in the urban and regional economic field, which argues that regional (and urban) units are increasingly relevant for the advancement of a country, as innovation leads increasingly rely on knowledge that tends to remain localised (Glaeser and Gottlieb (2009)).

New empirical research on knowledge spillovers, technological diffusion and regional innovation (Cortinovis and van Oort (2019); Miguélez and Moreno (2015); Caragliu and Nijkamp (2016)) also stress the importance of absorptive capacity of regions to address issues related to the impact of knowledge spillovers on innovation activities. Caragliu and Nijkamp (2016) analyse how various aspects of proximity (such as geography, cognitive, relational) impact on regional knowledge spillovers, and find that absorptive capacity is critical for a region to benefit from the knowledge produced in other regions. Miguélez and Moreno (2015) and Cortinovis and van Oort (2019) support these findings and stress the importance of networks and spatial mobility. They both emphasise that it takes a critical mass of existing knowledge to capture R&D spillovers. The formal model we propose borrows insights from this literature by incorporating disparities in the spatial extent of knowledge externalities, and migration costs, so that the region with the larger knowledge spillovers is characterised by higher productivity and a higher population share.

The present work is closely related to the literature on endogenous growth and endogenous industry location pioneered by Grossman and Helpman (1991) and subsequently adopted by the new economic geography literature (Martin and Ottaviano (1999), Baldwin and Forslid (2000); Martin and Ottaviano (2001); Baldwin et al. (2011)). Grossman and Helpman (1991) show that, when knowledge spillovers are global, initial conditions determine the pattern of trade and growth. In their approach, however, there is a single R&D sector that innovates and each innovation is used to produce a new variety. Hence, innovation takes place only in the country with the larger stock of knowledge capital. In our setup, in contrast, dispersion of innovation activities can be compatible with regions of different sizes and of, potentially, different knowledge endowments. Building on Grossman and Helpman (1991), Martin and Ottaviano (1999) consider the role of global and local R&D spillovers; they show that geography influences productivity only if spillovers are local and, then, study the effect of lower transport costs on agglomeration and growth. Rather than assuming one or the other type of spillover, we take into consideration both the strength and spatial extent of knowledge externalities, and show how regional disparities in innovation-enhancing activities, rather than transport frictions, can affect both industry location and growth. Both Baldwin and Forslid (2000); Martin and Ottaviano (2001), develop growth-and-geography models featuring monopolistic competition, endogenous industrial location and inter-regional migration, with the aim to analyse the spatial evolution of economic activities. Namely, the conditions that support multiple equilibria and the role played by impediments to trade and factor mobility. 6 Compared to this literature, an added feature of our approach is that the number of firms is endogenous, allowing us to relate industry concentration and growth. We show that reducing the knowledge spillover deficit across regions may lead to higher growth and more concentrated markets. Although the link between market power concentration, innovation and

⁶ Baldwin et al. (2011) provide a comprehensive overview of this literature.

In its basic structure, our model is closely related to the GOLE framework developed by Neary, in that we develop a two-region model (without any competitive numeraire sector) featuring a large number of variety of goods produced by differentiated sectors with firms competing à la Cournot (see, e.g., Eckel and Neary (2010); Neary (2016)). In our oligopolistic economy, though, there is endogenous growth and the focus is on how regional disparities in innovation activities affect the geography of economic activity and inter-regional inequality, rather than on the implications of international trade.

growth is well established in the literature (Aghion and Howitt (1992); Peretto (1996); Aghion et al. (1997); Etro (2009)), we explicitly consider location as an additional variable, which allows for the endogenous emergence of regional disparities in innovation clusters. In the paper we also touch upon the issue of scale distortions embedded in the NEG approach (Bond-Smith (2019)), and extend the model so that the spatial consequences of innovation are not amplified by scale effects.

Our present contribution also resonates with recent work on spatial economics incorporating trade costs and labour mobility, where part of the spatial variation in income across regions is explained by variations in trade costs (Allen and Arkolakis (2014); Desmet et al. (2018)). Differently from this literature, we focus on the endogenous emergence of industry clusters, growth and inter-regional inequality. Compared to this literature, our contribution is to show that spatial differences in knowledge externalities (coupled with migration costs) not only ensure uniqueness and existence of a spatial equilibrium but are relevant for industry location and growth. Crucially, we are able to explain dispersed innovation activity in the presence of no transportation costs.⁸

The paper is organised as follows. Section 2 describes the model. Section 3 analyses the equilibrium, first conditional on a given population distribution across regions, and then conditional on individuals' location choice. Section 4 analyses the effect of knowledge spillovers on the patterns of agglomeration, or dispersion, of activities and individuals. Section 5 explores the implications for growth and inter-regional income inequality. Section 6 briefly discusses the role of transport costs, while in Section 7 we look at the implications of relaxing firm homogeneity and fixed population. Section 8 concludes. All proofs are presented in the Appendix Table 1 below details the notation (parameters and variables) used throughout the paper.

2. The Model

We consider a two-region setup (North and South) where the same disembodied form of technology (e.g., blueprints, intangibles) may be adopted in the whole economy, but one region is better than the other at capturing outside (non-codified) knowledge. As a result, the region with the larger knowledge spillovers experiences lower innovation costs and, thereby, higher productivity. We assume that knowledge is embedded in labour hired and used in R&D activities. Workers are homogeneous and are used for both R&D as well as the production of goods. They are perfectly mobile within a region but imperfectly mobile between regions. In what follows, we restrict attention to the description of technologies and preferences in the Northern region. Analogous expressions apply to the Southern region. Whenever a distinction is needed variables and/or parameters for the South are denoted with a star, *.

Time t is continuous and goes from zero to infinity. The economy as a whole has a constant, exogenous number of identical, infinitely-lived, skilled workers, L^w , each endowed with one unit of labour-time supplied inelastically.

It is assumed that, at date zero, a share η of individuals are born and reside in the North, while the remaining share $\eta^* = 1 - \eta$ is born and reside in the South.

The labour market is perfectly competitive and workers incur a positive non-pecuniary cost of migration. The latter accounts, among other things, for the cost of adapting to a

⁸ Tabuchi et al. (2018) also have migration costs acting as the dispersion force. In their paper, however, the distribution of activities is determined by the interplay between labour productivity and migration costs; moreover, in their work, technological progress is exogenous and affects all regions equally. In contrast, we assume that firms' ability to capture knowledge differs across regions, and identify knowledge spillovers as an agglomeration force.

Table 1. Notation

Symbol	Description
N	Number of industrial sectors, $j = 1,, N$
Q_i	Number of firms in each sector $j = 1,, N$, each denoted by $q_j = 1,, Q_j$
α_i	Parameter of taste for each good $j, j = 1,, N$
ρ	Rate of time preference
η	Share of total population residing in the North
L^w	Total number of individuals/total quantity of labour (North+South)
$ au_D$	Intra-regional transportation cost
$ au_I$	Inter-regional transportation cost
μ	Degree of inter-regional knowledge spillover in R&D
E_t	Per-capita level of expenditures
$c_{i,t}$	Consumption of good $j, j = 1,, N$
$p_{j,t}$	Price of good $j, j = 1, \dots, N$
ν	Returns to knowledge in the R&D sector
δ	Productivity parameter in the R&D sector
$A_{q_i,t}$	Quantity of knowledge produced by a Northern firm q_i , $q_j = 1, \ldots, Q_j$
$X_{q_i,t}$	Quantity of good produced by firm q_j , $q_j = 1, \ldots, Q_j$
$L_{a:t}^{\widetilde{X}}$	Quantity of labour devoted to the production of good q_i , $q_i = 1, \dots, Q_i$
$X_{q_j,t} \ L_{q_j,t}^X \ L_{q_j,t}^A$	Quantity of labour devoted to R&D in firm q_i , $q_j = 1,, Q_j$
γ ,,,	Endogenous share of sectors located in the North
w_t	Wage rate in the Northern region
$\Pi_{q_i,0}$	Present values of expected profits of firm q_i , $q_j = 1, \dots, Q_j$
$\pi_{q_i,t}$	Time t profits of firm $q_j, q_j = 1, \dots, Q_j$
r_t	Real interest rate
m	Individual migration cost
ξ_t	Co-state variable associated with the R&D technology in every firm $q_j, q_j = 1,, Q_j, j = 1,, N$

new environment, moving away from friends and family, and similar. As made clear later on, such a migration cost allows a steady-state equilibrium in which individuals stay put while firms, responding to shocks, set up operations in one region or the other.

Preferences and technologies are described next. Individuals derive utility from the consumption of diverse goods with preferences given by

$$U = \int_0^\infty \left[\sum_{j=1}^N \alpha_j \log \left(c_{j,t} \right) \right] e^{-\rho t} dt \,, \tag{1}$$

where α_j is a parameter of the taste for variety j, with $\sum_{j=1}^N \alpha_j = 1$, N > 2 represents the exogenous set of commodities produced in the whole economy, $c_{j,t}$ is the consumption of variety j ($j = 1, \ldots, N$) and $\rho > 0$ is the rate of time preference. To simplify we impose $\alpha_j = \alpha = 1/N$. Lifetime utility, expression (1), is slightly different from that used in standard models in the new economic geography literature (NEG). In particular, there is no homogeneous (agricultural) sector good which can be traded at no cost between regions. The budget constraint of an individual residing in the North is given by

⁹ In NEG models this is the device used to equalise the wage of (unskilled) individuals in both regions. Another point of departure with the standard literature is the use of a Cobb-Douglas felicity function instead of a standard CES form. Such formalisation, however, is made to simplify the analysis. Detailed calculations, available from the authors upon request, show that none of the results we derive hinge on this specification.

$$E_{t} = \sum_{j=1}^{\gamma N} \tau_{D} p_{j,t} c_{j,t} + \sum_{j=\gamma N+1}^{N} \tau_{I} p_{j,t}^{*} \overline{c}_{j,t}, \qquad (2)$$

where E_t denotes the per-capita level of expenditure, $p_{j,t}$ ($p_{j,t}^*$) is the price of commodity j produced in the North (South), τ_D and τ_I stand for transport (iceberg) costs, and γ represents the (endogenous) share of industries located in the North. Accordingly, γN is the number of commodities produced in the North. The upper-bar indicates the consumption of a good produced in the foreign region (imported good).

The iceberg costs can also be interpreted as capturing the quality of infrastructure within a region, τ_D , or between regions, τ_I (see Martin (1999)). In line with the literature we impose the restriction $1 \le \tau_D \le \tau_D^* < \tau_I = \tau_I^*$, that is, transport costs are less costly within a region than between regions, and infrastructure in the North is of better quality than in the South. In each period t, every variety j of commodities (j = 1, ..., N) is produced by an endogenous number of identical firms $Q_{j,t} > 1$, each designated by q_j $(q_j = 1, ..., Q_{j,t})$, competing "à la Cournot". Both $Q_{j,t}$ and γ are crucial variables of the model as they are related to the extent of firms' market power and the extent of firms' agglomeration, respectively. To keep the analysis simple and in line with empirical evidence, we suppose that in each industry j, every firm q_j engages simultaneously in the production of good j and in R&D (see, Dasgupta and Stiglitz (1980)).

Denoting by $X_{q_j,t}$ the quantity of good produced by firm q_j , the total amount of variety j is $X_{j,t} = \sum_{q_{i-1}}^{Q_{j,t}} X_{q_j,t}$. The technology of production of every firm is given by

$$X_{q_{j},t} = (A_{q_{j},t})^{\nu} L_{q_{j},t}^{X}, \tag{3}$$

where $0 < \nu < 1$ denotes the returns to knowledge, $L_{qj,t}^X$ is the quantity of labour devoted to the production of a good, and $A_{qj,t}$ is the stock of (specific) knowledge-capital produced and used by firm q_j . Each firm can improve its productivity over time by engaging in inhouse R&D via a process of cost reduction driven by the accumulation of firm-specific knowledge-capital (cf. Peretto (1996)). The production function (3) does not include a fixed or sunk cost, only variable costs. However, notice that in the (infinite horizon) open-loop dynamic model we present, firms' R&D expenditures are effectively sunk at every point in time by all active firms. Hence, they are formally equivalent to fixed production or maintenance costs (Spence (1984)).

The number of units of knowledge-capital produced per unit of time by each firm q_j is given by

$$A_{q_{j},t}^{*} = \delta L_{q_{j},t}^{A} \left[\sum_{j=1}^{\gamma N} \sum_{q_{j}=1}^{Q_{j,t}} A_{q_{j}t} + \mu \sum_{j=\gamma N+1}^{N} \sum_{q_{j}=1}^{Q_{j,t}} A_{q_{j}t}^{*} \right], \tag{4}$$

where $\delta > 0$, is the R&D productivity parameter. The term $L_{q_j,t}^A$ is the amount of labour devoted to R&D, and μ denotes the degree of inter-regional knowledge spillover. Expression (4) represents the flow of knowledge generated by R&D. The R&D technology (4) above exhibits constant returns to scale in the factor that is accumulated, i.e knowledge. Within each region firms are able to take full advantage of each other's knowledge, also helped by intra-regional perfect mobility of workers; however, outside each region

Notice that there are no segmented markets in this economy. Specifically, we assume a world market for each commodity j = 1, ..., N, in which firms cannot price discriminate. Notice also, that we consider a symmetric Cournot equilibrium, which entails industrial specialisation within regions.

knowledge spillovers are not perfect and are region specific. Formally, this amounts to imposing the restriction: $0 \le \mu^* < \mu \le 1$.

To close the model, we set the labour constraint in the North as

$$\eta L^{w} = L_{\star}^{X} + L_{\star}^{A} \,, \tag{5}$$

where $L_t^X = \sum_{j=1}^{\gamma N} \sum_{q_j=1}^{Q_{j,t}} L_{q_j,t}^X$ and $L_t^A = \sum_{j=1}^{\gamma N} \sum_{q_j=1}^{Q_{j,t}} L_{q_j,t}^A$ denote the *aggregate* quantity of labour employed in the production of differentiated goods and employed in R&D, respectively. In the sequel since the total number of workers is fixed, without loss of generality, we set $L^w = 1$.

3. Equilibrium

We proceed in three steps. First, we describe the behaviour of individuals and firms. Second, we derive the equilibrium of the model assuming a given population distribution across regions, $\eta \in [1/2, 1]$, and no migration. Third, we analyse the choice of location of individuals, and study the spatial equilibrium. As regards market structure, we assume Cournot competition with free entry in the goods market (see below), and perfect competition in the labour market. We denote by w_t the price of labour in the North and normalise the price of labour in the South to one, that is $w_t^* = 1$.

3.1 Individuals and Firms

Each individual maximises lifetime utility (1) subject to the budget constraint (2). The solution of this programme is standard. The demand function for consumption good j of an individual living in the North is given by $c_{j,t} = E_t/(N\tau_D p_{j,t})$ if $0 < j \le \gamma N$ and $\overline{c}_{j,t} = E_t/(N\tau_I p_{j,t}^*)$ if $\gamma N < j \le N$. The aggregate demand function for variety j, denoted $c_{j,t}^d$, is thus given by,

$$c_{j,t}^{d} = \frac{1}{N} \left(\frac{\eta E_t}{\tau_D p_{j,t}} + \frac{(1 - \eta) E_t^*}{\tau_I p_{j,t}} \right).$$
 (6)

Firms perform two activities: (i) they produce and sell in an oligopolistic Cournot market with free entry and exit and, (ii) they generate new pieces of knowledge-capital via their in-house R&D using labour.

The market equilibrium we consider is a symmetric Nash equilibrium in open-loop strategies. Denote by $s_{q_i} = [X_{q_{it}}, L_{q_{it}}^A, A_{q_{it}}]$ for $t \ge 0$ firm q_j 's strategy vector. To make the

¹¹ It would be possible to also introduce imperfect intra-regional knowledge spillovers. In this case, the technology (4) would read: $A_{q_j,t} = \delta L_{q_j,t}^A[(1-\lambda)A_{q_j,t} + \lambda \sum_{j=1}^{\gamma N} \sum_{q_j=1}^{Q_{j,t}} A_{q_j,t} + \mu \sum_{j=\gamma N+1}^{N} \sum_{q_j=1}^{Q_{j,t}} A_{q_j,t}^*]$, where λ would stand for the degree of intra-regional knowledge spillover, and verify: $0 \le \mu^* < \mu < \lambda^* < \lambda \le 1$. However, imperfect intra-regional knowledge spillovers are not essential for our results and seriously complicate the model. The more general analysis of the model for $0 \le \mu^* < \mu < \lambda^* < \lambda \le 1$ is available from the authors upon request.

The advantage of focusing on an open-loop equilibrium is that it allows a closed-form solution. The drawback of using an open-loop equilibrium is that, typically, it does not have the property of subgame perfection. Unfortunately, closed loop or feedback equilibria complicate the model substantially, as they do not guarantee a closed form solution and, often, do not allow for a solution at all.

analysis simple, we assume that entry and exit involve zero costs, 13 meaning that the number of firms can freely adjust to its equilibrium level. In equilibrium firms commit to time paths for production, R&D, and labour at time t, with entry and exit determining the number of active firms, Q_j . Therefore, at time t the vector $[Q_j, s_1, \ldots, s_j, \ldots s_{Q_j}]$ is an instantaneous equilibrium with free entry and exit if for all firms q_j (and in all sectors, $j = 1, \ldots N$)

$$\Pi_{q_{i},t}[Q_{j},s_{1},\ldots,s_{j},\ldots s_{Q_{i}}] \ge \Pi_{q_{i},t}[Q_{j},s_{1},\ldots,s'_{j},\ldots s_{Q_{i}}] \ge 0,$$
(7)

and for $Q_i > 1$

$$\Pi_{q_i,t}[Q_j+1,s_1,\ldots,s_j,\ldots s_{Q_j+1}] \le 0,$$
 (8)

where $[Q_j, s_1, \ldots, s'_j, \ldots s_{Q_j}]$ is the strategy vector when firm q_j deviates from its optimal time-paths while all other firms do not. Condition (7) requires that a firm maximises the sum of present values of its net profits while taking as given the behaviour of the other firms, and this value be non-negative. Condition (8) is a standard zero-profit condition. Accordingly, each firm q_j (in the North) maximises

$$\Pi_{q_{j},t} = \int_{0}^{\infty} [p_{j,t} X_{q_{j},t} - w_{t} L_{q_{j},t}^{X} - w_{t} L_{q_{j},t}^{A}] e^{-\int_{0}^{t} r_{u} du} dt,$$
(9)

subject to equations (3) and (6), and taking as given the law of motion of knowledge-capital in the R&D sector (4) and the real interest rate, r_t . After substitution, the current value Hamiltonian becomes

$$CVH_{q_{j},t} = \left\{ \begin{aligned} X_{q_{j},t} \Big[\Omega_{t} \Big(\sum_{q_{j}=1}^{Q_{j,t}} X_{q_{j},t} \Big)^{-1} - w_{t} (A_{q_{j},t})^{-\nu} \Big] - w_{t} L_{q_{j},t}^{A} \\ + \xi_{t} \delta L_{q_{j},t}^{A} \Big[\sum_{j=1}^{\gamma N} \sum_{q_{j}=1}^{Q_{j,t}} A_{q_{j},t} + \mu \sum_{j=\gamma N+1}^{N} \sum_{q_{j}=1}^{Q_{j,t}} A_{q_{j},t}^{*} \Big] \right\}, \end{aligned}$$

where

$$\Omega_t \equiv \frac{1}{N} \left[\frac{\eta E_t}{\tau_D} + \frac{(1 - \eta) E_t^*}{\tau_I} \right], \tag{10}$$

and $\sum_{j=1}^{\gamma N} \sum_{q_j=1}^{Q_{j,t}} A_{q_j,t} + \mu \sum_{j=\gamma N+1}^{N} \sum_{q_j=1}^{Q_{j,t}} A_{q_j,t}^*$ are taken as given by each firm. The term ξ_t is the co-state variable associated with (4) and $\sum_{q_j=1}^{Q_{j,t}} X_{q_j,t} = c_{j,t}^d$. In this problem, the choice variables are: $X_{q_j,t}$ (production of commodity j), $L_{q_j,t}^A$ (quantity of labour employed in R&D) and $A_{q_j,t}$ (the path of knowledge-capital). The first order conditions are given by: $\partial CVH_{q_j,t}/\partial X_{q_j,t} = 0$, $\partial CVH_{q_j,t}/\partial L_{q_j,t}^A = 0$ and $\partial CVH_{q_j,t}/\partial A_{q_j,t} = -\xi_t^* + r_t\xi_t$. The transversality condition is $\lim_{t\to\infty} \xi_t A_{q_i,t} e^{-\int_0^t r_u du} = 0$.

Obviously this is a strong assumption. Effectively it is implying that R&D knowledge (as embodied in labour hired and used in R&D activities) is substitutable across firms and varieties. It is made in order to keep the model and its dynamics tractable. Notice though that R&D expenditure forms part of firm's total costs and is determined endogenously in market equilibrium, cf. (Peretto, 1996, p.897). Prospective entrants are aware that these costs have to be incurred in the post-entry equilibrium.

Note that, the only plausible Nash-equilibrium of the model is associated with $\partial CVH_{q_j,t}/\partial L_{q_j,t}^A=0$, at which the marginal revenue of an extra unit of labour devoted to R&D equals its cost (here w_t) and there is no incentive for firms to deviate from the strategy $L_{q_j,t}^A>0$.

Straightforward computations yield

$$X_{j,t} = \frac{\Omega_t}{w_t (A_{q_j,t})^{-\nu}} \left(1 - \frac{X_{q_j,t}}{X_{j,t}} \right), \tag{11}$$

$$w_{t} = \xi_{t} \delta \left(\sum_{j=1}^{\gamma N} \sum_{q_{j}=1}^{Q_{j,t}} A_{q_{j},t} + \mu \sum_{j=\gamma N+1}^{N} \sum_{q_{j}=1}^{Q_{j,t}} A_{q_{j},t}^{*} \right), \tag{12}$$

$$r_{t} = \nu (1 - X_{q_{j},t}/X_{j,t}) \left(\frac{w_{t} L_{q_{j},t}^{A}}{X_{q_{j},t}/X_{j,t}}\right) \frac{1}{\xi_{t} A_{q_{j},t}} + \frac{\xi_{t}^{\bullet}}{\xi_{t}}.$$
 (13)

Equation (11), where $X_{j,t} \equiv \sum_{q_j=1}^{Q_{j,t}} X_{q_j,t}$, is the total production of variety j, implicitly gives the best response of firm q_j (i.e., $X_{q_j,t}$) to the choice of production of good j of the other firms. Note that this condition is used to determine the price level of each variety.

Since $p_{j,t} = \Omega_t(c_{i,t}^d)^{-1} = \Omega_t(X_{j,t})^{-1}$ (see 6), we obtain,

$$p_{j,t} = \frac{w_t (A_{q_j,t})^{-\nu}}{(1 - X_{q,t}/X_{j,t})}.$$
 (14)

Equation (14) shows that the price of each variety is determined by the product between its marginal cost of production $\left(w_t(A_{q_j,t})^{-\nu}\right)$ and the markup $1/(1-X_{q_j,t}/X_{j,t})>1$. Equation (12) is a static condition equating the marginal cost (w_t) and benefit $(\xi_t\delta(\sum_{j=1}^{\gamma N}\sum_{q_j=1}^{Q_{j,t}}A_{q_j,t}+\mu\sum_{j=\gamma N+1}^{N}\sum_{q_j=1}^{Q_{j,t}}A_{q_j,t}^*))$ of an additional unit of labour spent in R&D. Finally, equation (13) is a dynamic condition stating that the return (r_t) of a new piece of knowledge-capital depends on three factors: the productivity gains from knowledge accumulation (cost reducing effect of R&D, (first term on the rhs)), the future units of knowledge-capital (second term on the rhs) and the change in the shadow price of knowledge-capital (third term on the rhs).

Using (9), we derive the standard condition $r_t\Pi_{q_j,t}=\Pi^{\bullet}_{q_j,t}+\pi_{q_j,t}$, where $\pi_{q_j,t}=p_{j,t}X_{q_j,t}-w_tL_{q_j,t}^X-w_tL_{q_j,t}^A$ is the instantaneous profit of a firm. Given that there is free entry and exit, we have $\Pi=\Pi^{\bullet}\leq 0$. Using (3) and (14), we then obtain,

¹⁴ If $\partial CVH_{q_j,t}/\partial L_{q_j,t}^A < 0$, the marginal revenue of an extra unit of labour devoted to R&D is always negative implying no labour allocated to R&D in equilibrium (i.e., $L_t^X = \sum_{j=1}^{\gamma N} \sum_{q_j=1}^{Q_{j,t}} L_{q_j,t}^X = \eta L^w$). Such an outcome can be ruled out, as any firm would have an incentive to deviate from the $L_{q_j,t}^A = 0$ strategy and choose $L_{q_j,t}^A > 0$ to produce new pieces of knowledge-capital, thereby improving their productivity and thus profitability vis-a-vis their rivals. If $\partial CVH_{q_j,t}/\partial L_{q_j,t}^A > 0$, the marginal revenue of an extra unit of labour allocated to R&D is always positive implying all labour allocated to R&D in equilibrium (i.e., $L_t^A = \sum_{j=1}^{\gamma N} \sum_{q_j=1}^{Q_{j,t}} L_{q_j,t}^A = \eta L^w$). The latter can also be ruled out, as it leads to a meaningless solution with no production of (differentiated) goods (cf. Peretto (1996)).

Note that, under perfect foresight, a firm can finance its own R&D through debt or equity, as the no arbitrage condition (13) implies that the rate of return from a risk-less loan must be equal to the cost of financing R&D by borrowing (cf. (Peretto, 1996, p. 904)).

$$\frac{\mathrm{pdw}_{t}L_{q_{j},t}^{X}}{(1-X_{q_{j},t}/X_{j,t})} - w_{t}L_{q_{j},t}^{X} - w_{t}L_{q_{j},t}^{A} \leq 0.$$
(15)

This is the zero profit condition for every firm $q_i = 1, \dots, Q_i$. Note that the standard asset-pricing equation collapses into a a zero-profit condition because the number of firms is a free to jump variable. ¹⁶

Finally, by combining (13), (3) and the zero profit condition above (15) we obtain the rate of return to R&D in the industry (partial) equilibrium, that is

$$r_{t} = \nu \left(1 - X_{q_{j},t} / X_{j,t}\right) \left(\frac{w_{t} L_{q_{j},t}^{A}}{X_{q_{j},t} / X_{j,t}}\right) \frac{1}{\xi_{t} A_{q_{j},t}} + \frac{\xi_{t}^{*}}{\xi_{t}}.$$
 (16)

Noticeably, the cost reducing effect of R&D can be decomposed into the return to R&D earned by the increase in the market share $\nu(1 - X_{q_i,t}/X_{i,t})$, and the returns to R&D earned for a given market share $\frac{w_i L_{q_j,l}^A}{X_{q_i,l}/X_{j,l}}$. The latter implies that the incentive to innovate is the incremental profit (gross-profit effect), while the former captures the potential gains from an increase in rivals' market shares (business-stealing effect). This is in keeping with many IO models of R&D and is in line with Peretto (1996). 17

3.2 Intra-regional equilibrium

Intra-regional equilibrium determines the number of firms in each sector, quantities and prices for given shares of population (η and $1-\eta$). We focus on an *intra*-regional symmetric equilibrium, whereby prices and quantities of goods are identical within a region but, as it will be the case, different between regions. Formally,

Definition 1. For all sectors j in a given region, an intra-regional symmetric equilibrium is characterised by the number of firms in each sector Q_{jb} quantities $X_{q_{jt}}$ and prices p_{jt} that are identical for all firms q_i . For the North, we have

- 1. $Q_{j,t} = Q_t$ for all $j \leq \gamma N$, 2. $X_{q_j,t} = X_{j,t}/Q_t = X_t/Q_t$, $L_{q_j,t}^X = L_{j,t}^X/Q_t = L_t^X/(\gamma NQ_t)$ $L_t^A/(\gamma NQ_t)$ for all q_j and all $j \leq \gamma N$, and and $L_{q_j,t}^A = L_{j,t}^A/Q_t =$
- 3. $p_{i,t} = p_t$ for all $j \leq \gamma N$. And similarly for the South.

Using the definition above, and combining the labour constraint (5) and the zero-profit condition (15), we obtain aggregate employment in the production of goods and R&D

¹⁶ This is a feature specific to the setting developed here and can be related to Peretto and Connolly (2007), who show that both horizontal and vertical dimension of technology are complementary. In their model, the quantity of resources required to perform horizontal innovation is decreasing due to some external effect that makes entry costs fall, and leads to closed-form solutions. In the limit case of zero entry cost, assumed here, it also implies that the transitional dynamics compresses to a jump to the new steady state (see (Peretto and Connolly,

¹⁷ In Peretto (1996) the return to R&D is non-monotonic in the number of firms as a result of the tension between the gross profit and the business stealing effects. Consequently, spillovers exert different influences on incentives to undertake R&D depending on which effect dominates. In our model, in contrast, the gross-profit effect always dominates the business stealing effect in (general) equilibrium (see Appendix 9.2). Recent empirical work by Bloom et al. (2013) supports our result.

$$L_t^X = \eta \left(1 - \frac{1}{Q_t} \right),\tag{17}$$

$$L_t^A = \eta \frac{1}{Q_t} \,. \tag{18}$$

That is, the share 1/Q of labour force is allocated to the production of R&D and, since $\eta \geq 1/2$ and $\mu > \mu^*$, the share of workers employed in each sector will be larger in the North (See proposition 1 below). Interpreting the number of firms in each sector as a proxy for the degree of markets' competitiveness $(Q = X_j/X_{q_j})$ it follows that: as firms' market share in a given sector increases $(Q \downarrow)$, firms allocate more labour to R&D and less to the production of commodities.

To ensure existence, and other results to follow, from now on we impose the following restriction on the returns to knowledge parameter, ν .

Assumption 1. Let $\nu > \frac{\rho}{n\delta}$

As shown in Appendix 9.1, Assumption 1 is a necessary and sufficient condition to ensure a strictly positive long-term growth rate and is easily satisfied for plausible parameter values.

In Appendix 9.1 we also formally demonstrate that a steady-state equilibrium exists, that the adjustment to steady state is instantaneous, and that growth of knowledge-capital is the same in the North and in the South. To simplify we assume that every firm starts with the same endowment of knowledge-capital $(A_{q_j,0} = A_{q_j,0}^*)$, though this is not essential for the results.¹⁸ The following proposition summarises the main properties of the steady state.

Proposition 1. (Steady state) For any given population distribution $\eta \in [1/2, 1]$

- a. There exists a unique steady-state equilibrium solution to which the economy jumps immediately,
- b. The steady state is characterised by a constant (and common to all firms) level of growth of capital-knowledge, $g = g^*$; and a constant and equal number of firms in each sector, $Q = Q^*$, where $Q > 1 + \frac{1}{12}$.
- c. $1/2 \leq \eta < \gamma$.

Proof. See Appendix 9.1

Regions grow at the same rate in steady state, but crucially relative productivity levels differ according to a region's ability to make the most of innovation. Also, the number of firms in each sector is the same for both regions in symmetric equilibrium (part b), and the more productive region has a higher share of sectors with $\gamma > \eta$ (part c). The latter is supported by ample empirical evidence (Carlino and Kerr (2014)) documenting that R&D activities are more concentrated than employment (the measure of agglomeration economies).

The equilibrium system expressed in terms of γ and Q (derived in Appendix 9.1) reads as

¹⁸ If $A_{q_j,0} \neq A_{q_j,0}^*$, the knowledge gap between firms of different regions would remain constant over time as the economy jumps immediately to the steady state (see Appendix 9.1).

$$(1-\eta)\left[1+\mu^*\frac{\gamma}{(1-\gamma)}\right] = \eta\left[1+\mu\frac{(1-\gamma)}{\gamma}\right],\tag{19}$$

$$Q = \frac{(1+\nu)[\gamma + \mu(1-\gamma)]\frac{\delta\eta}{\gamma}}{\nu[\gamma + \mu(1-\gamma)]\frac{\delta\eta}{\gamma} - \rho} > 1.$$
 (20)

From equation (19) we can express γ as a function of exogenous parameters, which then, through (20), gives Q as function of exogenous parameters.

Starting from the equilibrium share of sectors, γ , few comments are in order. First, there is a differentiated spillover effect according to which the share of industries located in each region is positively related to the region specific spillover (for details see Appendix 9.3). This effect is reinforced by a factor endowment effect (captured by the terms in η in (19)) according to which the share of sectors located in a given region is positively related to its population share: $d\gamma/d\eta > 0$ (readily follows from (19)). Intuitively, a larger population share creates a larger market for commodities and allows firms to conduct R&D at a greater scale. This is also an agglomerating force, and reminiscent of a home market effect whereby the larger region hosts a more than proportionate share of industries (see Krugman (1980)).

Turning to the equilibrium number of firms, Q, we can establish that $dQ/d\mu < 0$, $dQ/d\mu^* < 0$ and $dQ/d\eta < 0$. Intuitively, when μ^* and/or μ take greater values, the productivity of labour in R&D improves. As a result, the demand of labour in R&D increases thereby rising R&D costs. This is equivalent to a rise in the entry level of R&D, leading to a reduction in the number of operating firms. Lastly, a higher concentration of people in the North, ceteris paribus, reduces competitiveness ($dQ/d\eta < 0$). In fact, a larger pool of workers allows firms to conduct R&D at a larger scale, which pushes up L^A (relative to L^X) and increases the mark up over marginal costs. These steady state properties are reminiscent of Smulders and Van de Klundert (1995) and imply, as we shall see shortly, that market concentration triggered by inter-regional knowledge spillovers is conducive of growth.

The equilibrium wage and general price level are analysed next.

Total production of commodity j must be equal to the number of firms, Q, times quantity produced by each firm, (3); then, using the definition of Ω_t (see (10)) and combining it with (5), (15) and (17), we obtain the wage equation for the North,

$$w = \gamma \left[\frac{E}{\tau_D} + \frac{(1 - \eta)E^*}{\eta} \right].$$

Using the same method, we derive the wage for the South,

$$w^* = (1 - \gamma) \left[\frac{\eta}{(1 - \eta)} \frac{E}{\tau_I} + \frac{E^*}{\tau_D^*} \right].$$

Taking into account that the level of expenditure per capita is determined by the wage $(E=w \text{ and } E^*=w^*)$ and $w^*=1$ (by normalisation), the level of wage in the North, w, must adjust so that both conditions are simultaneously satisfied. After straightforward computations, we obtain the following expression

$$\frac{\eta}{(1-\eta)} \left[\frac{\eta}{(1-\eta)} w + \frac{\tau_I}{\tau_D^*} \right] = \frac{\gamma}{1-\gamma} \left[\frac{\eta}{(1-\eta)} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right],\tag{21}$$

implicitly defining w as function of all exogenous variables. Therefore, we establish the following proposition.

Proposition 2. (WAGE IN THE NORTH). For any given $\eta \in [1/2, 1]$, an equilibrium level w exists and is unique; moreover $w > w^* = 1$.

Proof. See Appendix 9.4. ■

In line with empirical evidence (Head and Mayer (2010); Redding (2011)) we find that the wage differential across regions is positively correlated with the population share in the larger region $(dw/d\eta > 0)$.²⁰ Intuitively, a larger share of workers implies a larger labour supply of labour that, in turn, prompts the location of a greater share of sectors $(d\gamma/d\eta > 0)$ and drives up the wage $(dw/d\gamma > 0)$. Note, also, that from (14) and $w^* = 1$ it immediately follows that $p/p^* = w$. Finally, transport costs are not crucial to explain the wage/price differential across regions. Indeed, $w > w^*$ for any $1/2 \le \eta < \gamma$ and no transport costs. ²¹

Turning to the general price index, hereafter denoted by P, we obtain 22

$$P = (\tau_D p)^{\gamma} (\tau_I p^*)^{1-\gamma} < P^* = (\tau_I p)^{\gamma} (\tau_D^* p^*)^{1-\gamma}, \text{ for } \eta \ge 1/2.$$
 (22)

In the region that hosts a larger share of industries, a greater proportion of goods are purchased without incurring the high inter-regional transport cost, τ_I . Therefore, individuals residing in the North benefit not only from a higher wage, but also from a lower price index than individuals located in the South; a set of circumstances that should make the Northern region more attractive. As we shall see shortly, this will play a role in the analysis of the migration decision of individuals.

Finally, we close this subsection by writing the common level of growth of know-ledge-capital in the North and South, that is,

$$g \equiv \frac{A_{q_j,t}}{A_{q_j,t}} = \delta L_{q_j,t}^A [\gamma NQ + \mu(1-\gamma)NQ].$$
 (23)

Accordingly, growth depends on labour employed in R&D and on the relative production of knowledge (term in squared brackets). In symmetric equilibrium $L_{q_j}^A = L^A/(\gamma NQ)$, and using (18) into the above we obtain,

$$g = \delta \eta \frac{1}{Q} \left[\frac{\gamma + \mu(1 - \gamma)}{\gamma} \right]. \tag{24}$$

Growth is determined by two endogenous variables: the number of firms per industry, Q, and the share of industries located in a given region, γ . Both Q and γ are variables affected by η , μ and μ^* , among other parameters. Later on in the paper, we carry out the equilibrium growth analysis and related comparative statics (see Section 5).

²⁰ Specifically, we obtain $dw/d\eta > 0$, with $\lim_{\eta \to 0} w = \mu \tau_D^* / \tau_I < 1$ and $\lim_{\eta \to 1} w = \tau_I / (\mu^* \tau_D) > 1$. See Appendix 9.4.

²¹ The role of transport costs is analysed in depth in Section 6.

²² The choice of a Fisher price index is made for convenience. Since it is computed as a geometric mean, it will be easier for us to interpret some of the results we derive later, particularly, regarding the migration condition.

3.3 Spatial equilibrium

Having set out the behaviour of agents, conditional on given shares of individuals in each region (η and $1 - \eta$), we now explore the conditions required for a spatial equilibrium. We begin by defining a spatial equilibrium.

Definition 2. A population share $\hat{\eta} \in (0, 1]$ is a spatial equilibrium, if individuals have no incentives to move away from the region in which they are originally located.

As long as the differential in utility between North and South (i.e. $\Delta U = U - U^*$) is lower, in absolute value, than the cost of migration, then, individuals born and residing in a given region stay put. Accordingly, using lifetime utility (1), we have,

$$\left| \begin{array}{l} \int_0^\infty [\sum_{j=1}^{\gamma N} \log{(c_{j,t})} + \sum_{j=\gamma N+1}^N \log{(\overline{c}_{j,t})}] e^{-\rho t} dt \\ - \int_0^\infty [\sum_{j=1}^{\gamma N} \log{(\overline{c}_{j,t}^*)} + \sum_{j=\gamma N+1}^N \log{(c_{j,t}^*)}] e^{-\rho t} dt \end{array} \right| \leq m,$$

where m denotes the migration cost of an individual.

Using the property that the economy jumps immediately to the steady state, together with the individual demand functions, and expressions (14) and (22), we can simplify the above to obtain,

$$|\Delta U| = \frac{1}{\rho} |\log w \frac{P^*}{P}| \le m. \tag{25}$$

Notice that, as w>1 and $P< P^*$, it follows that $\Delta U>0$ implying that migration never occurs from North to South. Indeed, in terms of utility, workers residing in the North fare better than those residing in the South, since they benefit from a higher wage and a lower price index. For later purposes, using (25) above and (22), the difference in the present value of utility reads as

$$\rho \Delta U = \log w + \log \left(\frac{\tau_I}{\tau_D}\right)^{\gamma} \left(\frac{\tau_D^*}{\tau_I}\right)^{1-\gamma},\tag{26}$$

and, differentiating (26) with respect to η gives,

$$\rho \frac{d\Delta U}{d\eta} = \frac{1}{w} \frac{dw}{d\eta} + \frac{d\gamma}{d\eta} \log \left(\frac{\tau_I}{\tau_D} \frac{\tau_I}{\tau_D^*} \right) > 0.$$
 (27)

Since $1 \le \tau_D \le \tau_D^* < \tau_I$, $d\gamma/d\eta > 0$ and $dw/d\eta > 0$, equation (27) implies that the utility differential is positively correlated with the population share in the North. In other words, a larger population in the North creates more incentives for individuals to migrate from the South to the North.

Keeping the above in mind, we can now turn to the characterisation of the spatial equilibrium, then to the analysis of migration and location of individuals and firms. In the spirit of Murata (2003), location preferences are idiosyncratic, implying a smooth migration function. Specifically, disregarding individuals in the North who always stay put, we

assume that the migration costs of individuals in the South, m, are drawn from a uniform distribution with support $[m^{low}, m^{high}]$. Depending on the size of m different scenarios emerge for individuals in the South. To analyse the possible outcomes, let us define the cut-off value \underline{m} such that $\Delta U - \underline{m} = 0$, that is the level of the migration cost at which individuals are indifferent between staying put or migrating

Solving $\Delta U = \underline{m}$ for the population share η , potentially, gives two possible solutions for the population distribution: one for the case of migration from North to South and the other for migration from South to North. As mentioned above, migration from the North to the South cannot occur, hence this solution can be ruled out. We thus denote by $\underline{\eta}$ the unique possible solution for the population distribution for which $\Delta U - \underline{m} = 0$. Notably, there is a one for one correspondence between \underline{m} and $\underline{\eta}$. This is because migration costs are drawn from the set of individuals in the South. Therefore, determining \underline{m} automatically sets η . Moreover, since $\Delta U(\eta) > 0$ and $d\Delta U/d\eta > 0$, it follows that $\eta > 0$.

If migration costs for individuals in the South verifies $m^{high} < \underline{m}$, and (25) always holds with a strict inequality (i.e., $\Delta U < m$ when $\Delta U - \underline{m} = 0$), then any initial population distribution is a dispersed spatial equilibrium. If, however, migration costs verify $0 < \underline{m} < m^{high}$, then any initial population distribution is an agglomerated spatial equilibrium. In these two extreme cases, migration costs are either low enough so that all individuals migrate to the North, or too large implying that individuals in the South stay put. Lastly, there is the intermediate case in which individuals in the South face migration costs verifying $0 < m^{low} < \underline{m} < m^{high}$. In this scenario, solving $\Delta U = \underline{m}$ determines the population distribution $\underline{\eta}$ that sustains a dispersed spatial equilibrium, with $0 < \eta < 1$. The following proposition summarises.

Proposition 3. (SPATIAL EQUILIBRIA). Define $\underline{\eta}$ as the population distribution satisfying $\Delta U - \underline{m} = 0$. Then there exists a critical cut-off value for the migration cost, \underline{m} , such that

- a. For $0 < m^{low} \le \underline{m} \le m^{high}$: Any initial population distribution $\eta_0 \in [0,1]$ is a dispersed spatial equilibrium, $\hat{\eta} = \eta$;
- b. For $0 < \underline{m} \le m^{low} < m^{high}$: Any initial population distribution $\eta_0 \in [0,1]$ is a dispersed spatial equilibrium, $\hat{\eta} = \eta_0$;
- c. For $0 < m^{low} < m^{high} \le \underline{m}$: Any initial population distribution $\eta_0 \in [0,1]$ is an agglomerated spatial equilibrium, $\hat{\eta} = 1$.

Proposition 3 establishes that, with migration, different types of spatial equilibria may arise. One is an agglomerated equilibrium: in part (c) individuals and firms locate in the North. In the remaining cases, we obtain a dispersed equilibrium where individuals and firms are located in both regions. Notably, while in part (b) some individuals initially born in the South have a migration cost which is too high to induce migration to the North, in the intermediate case (part a) any change in ΔU will affect the corresponding value of the migration cost \underline{m} (solution of $\Delta U = \underline{m}$) and, thereby, the corresponding population distribution, $\underline{\eta}$. In the latter scenario, individuals can potentially migrate to the North, provided that the utility differential between North and South diminishes, as will be discussed further in the following sections.

Finally, we link Proposition 3 with Proposition 1 that established conditions of existence and uniqueness for a potential dispersed equilibrium, requiring $\eta \in (1/2,1)$. Recall that in the context of Proposition 1, individuals were assumed to stay put and migration costs had not been introduced. To proceed, we need to compare $\underline{\eta}$ and 1/2. In principle, we can have $0 < \underline{\eta} < 1/2$ or $1/2 \le \underline{\eta} < 1$. Indeed, if $\underline{\eta}$ is close to 1/2 we can have

 $\Delta U(\underline{\eta}) > 0$, with $\underline{\eta} < 1/2$. However, this case can be ruled out as $\underline{\eta} < 1/2$ would contradict the condition of existence and uniqueness (Proposition 1). Hence, under Proposition 1, the solution to (25) necessarily implies $1/2 \leq \underline{\eta}$. Accordingly, we establish the following.

Corollary 1. For $0 < m^{low} \le \underline{m} < m^{high}$ and $\eta_0 \in (1/2, 1)$, there exists a unique spatial equilibrium $\hat{\eta} = \eta$ where individuals and firms are dispersed.

Corollary 1 is instrumental for the analysis carried out in the next section. Before proceeding further, it is instructive to study the parameter conditions under which the corollary applies. As shown in Appendix 9.4, $\lim_{\eta \to 1} w = \frac{1}{\mu^*} \frac{\tau_L}{\tau_D}$; thus, in the limit $\eta = 1$ the utility differential between North and South (25) reads as $|\Delta U| = \frac{1}{\rho} |\log \left[\frac{1}{\mu^*} \left(\frac{\tau_L}{\tau_D}\right)^2\right]|$. Since the term in square brackets is greater than one, the equation $\Delta U = m$ has a solution for m given by $m = \frac{1}{\rho} \log \left[\frac{1}{\mu^*} \left(\frac{\tau_L}{\tau_D}\right)^2\right]$. Accordingly, as long as $m^{low} \leq \frac{1}{\rho} \log \left[\frac{1}{\mu^*} \left(\frac{\tau_L}{\tau_D}\right)^2\right] < m^{high}$, Corollary 1 applies. Closer examination of this condition shows that a lower interregional transportation cost, τ_I , or a greater rate of time preference, ρ , or a larger knowledge spill-overs in the South, μ^* , or a higher intra-regional transportation cost in the North, τ_D , reinforce the conditions for a dispersed equilibrium. Intuitively, the parameters τ_I , τ_D and μ^* act through the relative wage and the relative price index (see 25) making the North less desirable. As for ρ , the potential future utility gains of migrating to the North are reduced as the rate of time preference increases.

4. Dispersion and Agglomeration

The previous sub-section has highlighted the role played by the non—pecuniary migration cost in establishing the kind of steady state we obtain. Here, we assess whether the interregional knowledge spillovers either reinforce the dispersed equilibrium or trigger a switch to an agglomerated equilibrium where individuals and firms agglomerate in a single (core) region. We carry out the analysis assuming that, initially, the economy is at a unique dispersed equilibrium (Corollary 1) and look at a decrease in μ^* for a given μ . As seen earlier, a greater gap in knowledge spillovers leads to a greater productivity in the R&D sectors of firms located in the North, thereby the share of industries operating in the North increases. This has also a positive impact on the wage, making the Northern region more attractive. Indeed, by differentiating (26) with respect to μ^* it can be easily checked that the utility differential between North and South expands as μ^* decreases, that is

$$\frac{d\Delta U}{d\mu^*} = \frac{1}{\rho} \left[\frac{1}{w} \frac{dw}{d\mu^*} + \frac{d\gamma}{d\mu^*} \log \left(\frac{\tau_I}{\tau_D} \frac{\tau_I}{\tau_D^*} \right) \right] < 0,$$
 (28)

making it more desirable for individuals to migrate. Thus, higher regional dispersion in knowledge spillovers may lead to agglomeration. However, it may be also compatible with a dispersed equilibrium, if differences in spillovers between regions are initially not too large, which is the most empirically plausible scenario.²³ In this case, the productivity

²³ The most recent empirical work, using U.S. firm level accounting data matched into U.S. Patent and Trademark Office data, documents that cross-regional spillovers are reduced in strength and correlated to geographical distance (Lychagin et al. (2016)).

advantage held by the most advanced region is not high enough to compensate for its higher wage triggered by higher demand of labour in the R&D sector. Notably, firms' dispersion is also robust to the possibility of migration. The following proposition summarises.

Proposition 4. (SPILLOVERS, DISPERSION AND AGGLOMERATION). Given the migration cost $0 < m^{low} < \underline{m} < m^{high}$ and the corresponding initial population distribution $\underline{\eta}$ that supports the dispersed spatial equilibrium, there exists a cut-off value $\overline{\mu}^*$ such that

- a. $\hat{\eta} = \eta$ is a dispersed spatial equilibrium for all $\overline{\mu}^* < \mu^* < \mu$, and
- b. $\hat{\eta} = \overline{1}$ is an agglomerated spatial equilibrium for all $\mu^* < \overline{\mu}^* < \mu$.

Proof. See Appendix 9.5. ■

This Proposition is important as it highlights the role played by the degree of spill-over in the South on individuals' decision to migrate and on firms' choice to relocate. To gain intuition, consider the two polar cases of μ^* decreasing and increasing, respectively. First, as μ^* decreases, individuals in the South will migrate to the North by increasing order migration costs. This will also induce firms to relocate to the North. In this case, we have both migration of individuals and relocation of firms. If, on the other hand, we consider an increase in μ^* , the North becomes less attractive. So individuals in the South stay put, meaning that the population distribution $\underline{\eta}$ that supports the initial equilibrium remains unchanged: there is no migration of individuals. However, we obtain a relocation of some sector to the South.

5. Growth and inter-regional inequality

To study the implications for growth and inter-regional income inequality, we focus on the case where the economy is at an equilibrium in which not all R&D activities and not all workers are concentrated in the most productive region (Corollary 1). Evidence suggests that this is indeed the most plausible real world scenario.

Using $L_{q_j}^A = L_j^A/Q = L^A/(\gamma NQ_t)$, (18) and plugging in equation (23) the value of Q given in Appendix 9.1 (see equation 20), we obtain the equilibrium growth rate,

$$g = \frac{1}{1+\nu} \left\{ \nu \frac{\delta \eta}{\gamma} [\gamma + \mu(1-\gamma)] - \rho \right\}. \tag{29}$$

Recall that in equilibrium the share of sectors γ depends on η , μ and μ^* (see (19)). Therefore, the equilibrium growth rate depends on the share of population located in the North and on knowledge spillovers. Tedious computations, relegated in appendix, lead to the following.

Proposition 5. (GROWTH). Along the equilibrium path defined by Corollary 1, and for any $\overline{\mu}^* < \mu^* < \mu$

a.
$$\frac{dg}{d\mu^*} > 0$$
 and $\frac{dg}{d\mu} > 0$;
b. $\frac{dg}{du^*} > \frac{dg}{du}$.

Proof. See Appendix 9.3.4. ■

Intuitively, productivity of R&D increases as knowledge spillovers increase (part a). Notably, this effect is stronger in the lagging region (part b). Recall that, as the R&D sector becomes more productive two things happen: (i) firms allocate a greater amount of labour to R&D and, (ii) more industries choose to operate in the region experiencing productivity gains. Since labour cost is higher in the North $(w > w^*)$, more industries set up in the South when μ^* increases than in the North for an equivalent increase in μ . As a result, μ^* has a stronger effect on growth than μ .

Turning to inter-regional income inequality, in the context of the present model, this is given by the wage gap between North and South. Recall that $w^* = 1$ and w comes from expression (21). The latter, is indirectly affected by μ and μ^* through γ .

Proposition 6. (SPILLOVERS AND INTER-REGIONAL INEQUALITY). Along the equilibrium path defined by Corollary 1, and for any $\overline{\mu}^* < \mu^* < \mu$, inter-regional inequality increases (decreases) with μ (μ^*).

Proof. See Appendix 9.6. ■

These findings suggest that strengthening the ability of the lagging region to capture knowledge leads to higher overall growth and a reduction of regional disparities. This is consistent, for example with empirical research documenting local spatial externalities between university research and high technology innovative activity, and the idea that promoting institutions that facilitate knowledge flows are important in supporting regional development (Jaffe (1989); Acs et al. (1992); Anselin et al. (1997); Kantor and Whalley (2014)).

From a welfare stand point, along the dispersed equilibrium path (and for any $\overline{\mu}^* < \mu^* < \mu$), we have seen that improving knowledge diffusion in the South reduces the welfare gap between North and South (see expression (28) establishing that $\frac{d\Delta U}{d\mu^*} < 0$). The latter, however, may come at the cost of creating regional winners and losers. Interestingly, we find that individuals in both regions are likely to gain. Computations (relegated in Appendix 9.7), show that the welfare effects of μ^* in the North and in the South amount, respectively, to

$$\frac{dU}{d\mu^*} = \frac{1}{\rho} \left[\frac{1/Q^2}{1 - 1/Q} \frac{dQ}{d\mu^*} + \frac{1 - \gamma}{w} \frac{dw}{d\mu^*} - \frac{d\gamma}{d\mu^*} \log w + \frac{d\gamma}{d\mu^*} \left(\log \frac{1}{\tau_D} - \log \frac{1}{\tau_I} \right) + \frac{\nu}{\rho} \frac{dg}{d\mu^*} \right]$$
(30)

and

$$\frac{dU^*}{d\mu^*} = \frac{1}{\rho} \left[\frac{1/Q^2}{1 - 1/Q} \frac{dQ}{d\mu^*} - \frac{\gamma}{w} \frac{dw}{d\mu^*} - \frac{d\gamma}{d\mu^*} \log w + \frac{d\gamma}{d\mu^*} \left(\log \frac{1}{\tau_I} - \log \frac{1}{\tau_D^*} \right) + \frac{\nu}{\rho} \frac{dg}{d\mu^*} \right]. \tag{31}$$

In the expressions above we can identify several welfare effects. The first term in squared brackets captures the degree of competition (static) effect, which is common to both regions and negative in sign. As seen earlier (Section 3.2), when μ^* increases, the number of firms in each sector decreases ($\frac{dQ}{d\mu^*} < 0$), reducing competitiveness and increasing the price of each good produced. The second term in squared brackets captures the change in the relative wage, and is negative for the North and positive for the South ($\frac{dw}{d\mu^*} < 0$); while the third and fourth terms capture the impact of the change in the share of industries operating in the North ($\frac{d\gamma}{d\mu^*} < 0$). The latter, noticeably, affects the relative

price index between regions in opposite directions, positively for the South and negatively for the North. Finally the fifth term, common to both regions, captures the long-term growth effect induced by greater knowledge spillovers in R&D. By differentiating (24) we obtain $\frac{dg}{d\mu^*} = -\frac{g}{Q}\frac{dQ}{d\mu^*} - \delta\eta \frac{1}{Q}\frac{\mu}{\gamma^2}\frac{d\gamma}{d\mu^*} > 0$, suggesting that the dynamic welfare gains are two-fold: (i) associated to the trade off between firms' market power and growth $(-\frac{dQ}{d\mu^*} > 0)$, and (ii) related to the effect of knowledge spillovers on the share of industries operating in the North $(-\frac{d\gamma}{d\mu^*} > 0)$. Notice, in particular, that the effect of knowledge spillovers on firms' mark up arises because the number of operating firms decreases as R&D becomes more productive; we term this the dynamic competition effect.

Tedious computations (relegated in Appendix 9.7) show that the dynamic competition effect dominates the static competition effect, i.e. $-\frac{g}{Q}\frac{dQ}{d\mu^*}>\frac{1/Q^2}{1-1/Q}\frac{dQ}{d\mu^*}$. In other words, the growth gains associated with fewer and more productive firms outweigh the static losses associated with a lower degree of competitiveness (higher prices). As a result, individuals in the South unambiguously gain from higher knowledge spillover in their region $(\frac{dU}{d\mu^*}>0)$. If, in addition, $\frac{dU}{d\mu^*}>0$, then individuals in the North also gain from narrowing the gap in knowledge spillovers between the advanced and the lagging region. By direct inspection of (30) and (31), and disregarding the term with transport costs, it is apparent that the change in the relative wage separates the two expressions. Hence for $\frac{dU}{d\mu^*}>0$ to hold it suffices to show that $\frac{1-\gamma}{w}\frac{dw}{d\mu^*}\leq \log w\frac{d\gamma}{d\mu^*}+\frac{1/Q^2}{1-1/Q}\frac{dQ}{d\mu^*}$, where $\frac{dQ}{d\mu^*}=\frac{\partial Q}{\partial \gamma}\frac{d\gamma}{d\mu^*}$. The following proposition restates the result.

Proposition 7. (Welfare). Assume transport costs are negligible $(\tau_D^* = \tau_D = \tau_I \to 1)$. Then along the equilibrium path defined by Corollary 1, and for any $\overline{\mu}^* < \mu^* < \mu$, narrowing the gap in knowledge spillovers result in welfare gains for both regions, if $\nu \leq \hat{\nu}$ where $\hat{\nu}$ is implicitly given by the solution of

$$\frac{\gamma \rho (1-\gamma)(1+\hat{\nu})[\gamma+\mu(1-\gamma)]}{\hat{\nu}(1-\eta)\mu\{\hat{\nu}[\gamma+\mu(1-\gamma)]\delta\eta-\gamma\rho\}} = 1$$

and γ is at its equilibrium level, given by equation (A.5) in the appendix.

Proof. See Appendix 9.7 ■

Since γ is independent of the parameters ν , δ and ρ the conditions ensuring $\nu \leq \hat{\nu}$ are not particularly restrictive; for instance, δ large enough relative to ρ would suffice. Proposition 7 implies that, transport costs aside, the welfare gains associated with the increase in the share of industries operating in the South are likely to outweigh the welfare losses associated with the lower relative wage in the North, implying a net welfare gain for the North too. Allowing for inter-regional transport costs weakens the potential welfare gains in the North, since it generates a larger wage gap across regions. This is explained in more detail in the following section.

6. Transport costs

In this section we briefly mention what role transport costs play in our model. In the economic geography literature it is usually argued that transport costs are crucial in shaping the distribution of activities (Krugman (1991); Fujita et al. (1999)) and in explaining the dynamics between growth and agglomeration (see, e.g, Minerva and Ottaviano (2009)). In these models, which typically assume monopolistic competition and increasing returns to

scale, the innovation sector requires goods which incur transport costs, so that industrial concentration, by reducing the input cost of innovation, increases the growth rate.

From direct inspection of (29) one can immediately check that, in our setup, the equilibrium growth rate does not depend on transport costs. This is because inter- and intra-regional transport costs $(\tau_I, \tau_D, \tau_D^*)$ do not directly influence the input costs of innovation and, therefore, play no role in the equilibrium location of industries (nor on their degree of competitiveness).

Transport costs, though, do influence the income differential between regions. Consider, for instance, the effect of a higher inter-regional transport cost. Simple algebra shows that w (see (21)) is increasing in τ_I if the number of industries located in the North is greater than in the South (i.e. $dw/d\tau_I > 0$ as $1/2 \le \eta < \gamma$). The reason is that an increase in τ_I induces a decrease in individual demands for foreign varieties and, as the Northern region is more populated in terms of firms ($\gamma > 1/2$), such an effect is more pronounced in the South than in the North. As a result, the relative price of varieties increases, which, ultimately, generates a larger wage gap across regions.²⁴ This also suggests that, ceteris paribus, the lower the transport cost the more likely are dispersed equilibria. Intuitively, when the transport cost τ_I decreases, on the one hand foreign varieties are cheaper and, on the other hand, the wage in the North decreases ($dw/d\tau_I > 0$). Both effects make the Northern region less attractive for individuals in the South.²⁵

Finally, due to the price index effect highlighted above, transport costs affect the change in regional welfare associated with higher μ^* . Namely, transport costs strengthen the positive welfare effect in the South but weaken the potential welfare gain in the North.

7. Extensions

In what follows, we discuss how our model may be extended to allow for population growth and firm heterogeneity. We draw from Peretto (1998) to explore the role of population growth, and from Impullitti and Licandro (2018) to introduce firm heterogeneity in our set-up. These extensions require some modifications and additional assumptions which are explained next. The more technical and lengthy computations are relegated in Appendix 9.9.

In line with Peretto (1998), we assume that population is growing exogenously at rate ω ; hence, $\frac{L_t^w}{L_t^w} = \frac{L_t}{L_t} = \frac{L_t^{**}}{L_t^*} = \omega$, with $0 < \omega < \rho$. For simplicity, the population growth rate is set identical across regions. The allocation of the world population remains: $L_t = \eta L_t^w$ and $L_t^* = (1 - \eta)L_t^w$ at any time. To account for population growth, the utility function N_t

now reads as
$$U = \int_0^\infty L_0[\sum_{j=1}^{N_t} \log{(c_{j,t})}/N_t]e^{-(\rho-\omega)t}dt$$
, where L_0 is the total number of indi-

viduals residing in the North at time zero, and $\rho - \omega$ can be interpreted as the net rate of time preference. Finally, to ensure constant growth rates of variables in steady state, the number of varieties grows exogenously at the same rate as population: $N_t = N_0 e^{\omega t}$, with N_0 representing the number of sectors at date zero. Hence, $N_t^* = \omega N_t$.

Incumbent firms' production technology (3) is unchanged, while the R&D technology is slightly modified. Following Impullitti and Licandro (2018), we assume:

The same logic applies to a change in τ_D and τ_D^* .

²⁵ Allen and Arkolakis (2014) and Tabuchi et al. (2018), also obtain that a lower inter-regional transport cost does not necessarily lead to agglomeration, while Martin and Ottaviano (1999) point out that transport costs are inconsequential for growth if spillovers are global rather than local.

$$A_{q_j,t}^{\bullet} = \delta L_{q_i,t}^A K_t,$$

where K_t is an external factor (knowledge spillover) defined as follow:

$$K_t = \overline{A}_{i,t}D_t,$$

and $\overline{A}_{j,t} = \sum_{q_j=1}^{Q_{j,t}} A_{q_j,t}/Q$ is the average productivity of those producing in the same sector j.²⁶ The term $D_t = \tilde{A}_t/\overline{A}_{j,t}$ denotes the difficulty of innovation, and is related to the distance between the average productivity of the overall economy and the average productivity of firms belonging to the same sector. Average productivity of the overall economy is given by: ²⁷

$$\tilde{A}_t = \frac{1}{N_t} \left[\sum_{j=1}^{\gamma N_t} (\overline{A}_{j,t}) + \mu \sum_{j=\gamma N_t+1}^{N_t} (\overline{A}_{j,t}^*) \right].$$

This specification implies that innovation is harder for firms belonging to sectors which use a greater amount of knowledge in their production process. ²⁸

Two comments are in order. First, compared to technology (4), the production of new pieces of knowledge which spill over is an average stock across sectors, rather than an entire stock. As it will become clear later, this feature is instrumental to avoid explosive growth in steady state. Furthermore, income per capita growth is now positively related to the average firm size, thereby eliminating scale effects (see, Laincz and Peretto (2006) and Bond-Smith (2019)).

Second, firms are heterogeneous. In line with Impullitti and Licandro (2018), heterogeneity between firms stems from the amount of knowledge required to produce a given variety of good j. Specifically, firms belonging to a given sector are identical, while firm productivity differs across sectors. Notation-wise, we have: $A_{q_i,t} = A_{j,t}/Q = \overline{A}_{j,t}$ for any $q_i = 1, \dots, Q$, and $A_{i,t} \neq A_{k,t}$ for any $j, k = 1, \dots, N_t$.

With regard to potential entrants, building on Impullitti and Licandro (2018), we assume that, at any time, a new variety among the non-operative varieties can be introduced at zero cost by Q firms associated to it. At the time of entry, each firm producing a new variety draws a common stock of knowledge, $A_{q_j,t} = A_{j,t}/Q = \overline{A}_{j,t}$, from a time-invariant distribution. Following Klette and Kortum (2004), we assume a discrete uniform distribution with support (A_t^{\min}, A_t^{\max}) for the North, and support $(A_t^{*\min}, A_t^{*\max})$ for the South, respectively.²⁹ We posit that the northern region is more productive implying that, at any moment, the average stock of knowledge in the North is greater than in the South, i.e., $\overline{A}_t = \sum_{j=1}^{\gamma N_t} (\overline{A}_{j,t}) / (\gamma N_t) > \overline{A}_t^* = \sum_{j=1}^{(1-\gamma)N_t} (\overline{A}_{j,t}^*) / [(1-\gamma)N_t].$

Finally, to ensure a steady state in which every firm (incumbents and entrants) grows at the same common growth rate, the distributions of knowledge must de facto depend on the endogenous growth rate, g. Formally, this means that, at any time t, in any sector j, northern (southern) firms draw a stock of knowledge given by: $A_{q_i,t} = \overline{A}_{j,0}e^{gt}$ $(A_{q_i,t}^* = \overline{A}_{i,0}^*e^{gt})$.

As firms are identical within a sector, we have $\overline{A}_{j,t} = A_{q_j,t}$ for any $q_j = 1, \dots, Q$. For the South, the amount of knowledge spillover is defined as: $A_t = \frac{1}{N_t} [\mu^* \sum_{j=1}^{\gamma N} (\overline{A}_{j,t}) + \sum_{j=\gamma N+1}^{N} (\overline{A}_{j,t}^*)]$. Impullitti and Licandro (2018) also assume decreasing returns to innovation (see p.195, and empirical evidence cited therein).

Akcigit and Kerr (2018) also extend the analysis by Klette and Kortum (2004), by developing a model in which heterogeneous firms perform both internal and external R&D. Their setup provides a characterisation of innovation behaviour across different-sized firms and its implications for growth. Here we stick to a simpler formulation to focus on the implications for the geography of knowledge of firm-specific R&D.

As noted by Impullitti and Licandro (2018), this assumption is akin to a spillover from incumbents to entrants. Furthermore, since distributions are time-invariant, it follows that $\overline{A}_t = \overline{A}_0 e^{gt}$ and $\overline{A}_t^* = \overline{A}_0^* e^{gt}$ at any time, where $\overline{A}_0 > \overline{A}_0^* > 0$ are the average productivity levels at date zero in the North and South, respectively. It, therefore, follows that firm heterogeneity translates into differences in average knowledge between regions. Repeating the same analysis of Section 3, we solve for the intra-regional equilibrium under heterogeneity and population growth. As shown in Appendix 9.9, we recover a set of equations reminiscent of the model with homogeneous firms, but for the terms ω and $\overline{A}_t/\overline{A}_t^* \equiv \Theta$. In the spirit of conciseness, in what follows we focus on the impact of heterogeneity on our main results; then, on the role of demographic shocks on the model outcomes.³⁰

Firm heterogeneity strengthens the productivity advantage of the North and, consequently, it weakens the relocation of firms from North to South when the degree of spillovers in the South increase, i.e., $\frac{d^2\gamma}{d\mu^*d\Theta} < 0$. Similarly, we find that the the impact on interregional utility differential is weakened, i.e., $\frac{d^2\Delta U}{d\mu^*d\Theta} < 0$. This is because heterogeneity mitigates the change in the relative wage between North and South and, thereby, the utility gain associated with higher knowledge spillover in the South. In conclusion, firm heterogeneity translates into differences in average knowledge between regions, mitigating the scale of firm relocation and inter-regional income and utility changes associated with a reduction in the gap in knowledge spillovers between regions.³¹

Finally, let us analyse the role of the rate of population growth. By direct inspection of (A.11) it can be checked that changes in the population growth rate have no effect on γ , since ω is common to both regions. For the same reason, the equilibrium wage is independent of ω ; therefore, changes in the rate of growth of population have no impact on the spatial equilibrium. On the other hand, higher population growth allows for a greater share of the workforce to be engaged in R&D. As a consequence, the number of firms in each sector decreases and the growth rate increases.

8. Concluding Remarks

In this paper, we have constructed a two-region growth model in which firms operate under Cournot competition and innovate through in-house R&D. Drawing from the evidence that both intra and inter-regional spillovers matter, our main aim was to study how regional disparities in innovation enhancing activities, coupled with migration costs, shape the geography of economic activities and the market structure.

We have shown that disparities in knowledge spillovers between regions lead to spatial concentration of industries, and the latter is associated with fewer, but more innovative

³⁰ See Appendix 9.9 for details on analytical computations.

³¹ Note that the effect of Θ on $dg/d\mu^*$ (and $dQ/d\mu^*$) is neutral. This is because the assumption of increasing innovation difficulty, which equalises growth rates, offsets the positive effect of greater productivity on innovation efforts.

³² Recall that population growth enters in the utility function, and the net rate of time preference is now given by ρ – ω < ρ. Thus, as population growth increases, individuals put a higher weight on future consumption relative to present consumption. This means that a lower amount of labour is allocated to production of commodities and a greater amount is allocated to R&D, leading to higher growth. Note that, in contrast to our model, in Peretto (1998) firms operate under monopolistic competition (implying that there is one firm per sector as if Q = 1), and since the number of firms grows at the same rate as population, productivity growth is independent of the population growth rate. In our model, with Cournot competition within each sector, productivity growth is affected by population growth because firms' market share in a given sector (1/Q) is itself affected by population growth.</p>

firms. Frictions in the movement of workers, on the other hand, limit the geographic concentration of economic activities in the most productive region. In this context, a weakening of the spatial disparities in knowledge spillovers between the advanced and the lagging region reduces income inequality, while preserving the positive effect on growth. This occurs because reductions in productivity advantages make firms more sensitive to wage differentials, leading to a rise in the share of industries operating in the lagging region. Welfare increases because the growth gains associated with fewer and more innovative firms outweigh the static losses associated with higher mark ups, and because the gains associated with the change in the spatial concentration of industries are likely to outweigh the losses associated with lower relative wages in the advanced region.

We have also considered whether transport costs play any role in shaping the distribution of activities across regions and found that the latter do not influence the equilibrium location of industries. This is consistent with the growing body of literature emphasising frictions in knowledge flows as source of agglomeration economies. Finally, extending the model to consider heterogeneous firms and population growth shows the robustness of our results.

9. Appendix

9.1 Proof of Proposition 1

Equation (19) draws from $g = g^*$ (Proposition 1 b) together with (4), (5) and (16), while equation (20) draws from $Q = Q^*$ (Proposition 1 b) together with (4), (12), (17) and (18) and the fact that, at the steady state, $r_t = \rho$. The proof is structured as follows. First we show that, in a *symmetric* equilibrium, the economy jumps immediately to a steady state where $g = g^*$. Second, that in the symmetric equilibrium, $Q = Q^*$. Finally we prove existence and uniqueness.

9.1.1 Step 1, showing that $g = g^*$

Using (3), (11) and (17), recall that the wage equation in the North is given by $\frac{\eta}{\gamma}w_t = \left(\frac{\eta E_t}{\tau_D} + \frac{(1-\eta)E_t^*}{\tau_I}\right)$ and an equivalent condition is satisfied for the South. Since $E_t^* = w_t^* = 1$ at each instant, and the consumer problem implies that $E_t^{\bullet}/E_t = E_t^{\bullet \bullet}/E_t^* = r_t - \rho$ at each instant, it follows that $E_t = E$ and $r_t = \rho$. Thus, the wage equations in the North and in the South imply that the wage in the North, and the share of sectors locating in the North (South) jump immediately to their steady state values.

From the individual demand functions $(c_t = E_t/(N\tau_D p_t))$ and $\overline{c}_t = E_t/(N\tau_I p_t^*)$ for $0 < j \le \gamma N$), it follows that $p_t c_t$ and $p_t^* \overline{c}_t$ must be constant. Therefore, at the aggregate level, $p_t c_t^d = p_t X_t$ is also constant. Since every firm in a given sector is identical, then $p_t X_t/Q_t$ $(p_t^* X_t^*/Q_t^*)$ must be constant. Therefore, the number of firms $Q(Q^*)$ in each sector $j = 1, \ldots, N$, jumps immediately to its steady-state value.

To complete the proof, we derive the dynamic equation. From equation (12), we have

$$w = \xi_t \delta \frac{A_{q,t}}{\gamma NQ} \left[\gamma NQ + \mu (1 - \gamma) NQ^* \frac{A_{q,t}^*}{A_{q,t}} \right]. \tag{A.1}$$

Differentiating with respect to time yields

$$-\frac{\xi_t^{\:\raisebox{3.5pt}{\text{\circle*{1.5}}}}}{\xi_t} = \frac{\delta\eta}{\gamma NQ^2} \left[\gamma NQ + \mu(1-\gamma)NQ^* \frac{1}{\hat{A}_t} \right] - \frac{\mu(1-\gamma)NQ^*}{\hat{A}_t [\gamma NQ + \mu(1-\gamma)NQ^*/\hat{A}_t]} \frac{\hat{A}_t^{\:\raisebox{3.5pt}{\text{\circle*{1.5}}}}}{\hat{A}_t} \:,$$

where we have denoted $\hat{A}_t = A_{q_j,t}/A_{q_i,t}^*$. Then, using (13), (12), (17) and (18), we obtain

$$-\frac{\xi_t^{\:\raisebox{3.5pt}{\text{\circle*{1.5}}}}}{\xi_t} = -\rho + \delta\nu\eta \frac{Q-1}{\gamma NQ^2} \left[\gamma NQ + \mu(1-\gamma) NQ^* \frac{1}{\hat{A}_t} \right] + \frac{\delta\eta}{\gamma NQ^2} \:.$$

Combining the two previous equations yields

$$\frac{\hat{A}_t}{\hat{A}_t} = \frac{\frac{\delta \eta}{\gamma N Q^2} \left[\gamma N Q + \mu (1 - \gamma) N Q^* \frac{1}{\hat{A}_t} \right] [1 - \nu (Q - 1)] + \rho - \frac{\delta \eta}{\gamma N Q^2}}{\frac{\mu (1 - \gamma) N Q^*}{\hat{A}_t [\gamma N Q + \mu (1 - \gamma) N Q^* / \hat{A}_t]}}.$$

Taking a first order Taylor approximation of the dynamic equation around the steady state, we obtain

$$\frac{\hat{A}_t}{\hat{A}_t} = -\frac{\frac{[1-\nu(Q-1)]\delta\eta}{\gamma Q} \mu(1-\gamma) \frac{\hat{A}_t - \hat{A}}{\hat{A}^2}}{\frac{\mu(1-\gamma)NQ^*}{\hat{A}_t[\gamma NQ + \mu(1-\gamma)NQ^*/\hat{A}_t]}},$$

where we have dropped the subscript t to indicate the steady-state value of \hat{A} . Direct inspection of this dynamic equation shows that, if $1 - \nu(Q - 1) < 0$ (i.e., $Q > 1/\nu + 1$) the economy jumps immediately to its steady state where $g = g^*$.

9.1.2 Step 2, showing that $Q = Q^*$

Using the fact that the growth rates are the same in both regions, together with (4), (5) and (15), we obtain

$$0 = \frac{\eta}{\gamma} [\gamma NQ + \mu(1 - \gamma)NQ^*] - \frac{(1 - \eta)}{(1 - \gamma)} [(1 - \gamma)NQ^* + \mu^* \gamma NQ].$$
 (A.2)

Using (3), (4), (12), (13), (17) and (18), we obtain

$$\rho = \delta\nu[\gamma NQ + \mu(1-\gamma)NQ^*] \frac{\eta(Q-1)}{\gamma NQ^2} - \delta[\gamma NQ + \mu(1-\gamma)NQ^*] \frac{\eta}{\gamma NQ^2}. \tag{A.3}$$

Noting that we can derive an equivalent expression for the South, the two equations above imply that $Q = Q^*$.

9.1.3 Step 3, Existence and uniqueness

From equation (A.2), using $O = O^*$, we obtain

$$\eta \left[1 + \mu \frac{(1-\gamma)}{\gamma} \right] = (1-\eta) \left[1 + \mu^* \frac{\gamma}{(1-\gamma)} \right]. \tag{A.4}$$

It can be easily checked that the left hand side (hereafter LHS) is strictly decreasing, with $\lim_{\gamma \to 0} LHS = +\infty$ and $\lim_{\gamma \to 1} LHS = \eta$. Similarly, the right hand side (hereafter, RHS) is

strictly increasing, with $\lim_{\gamma \to 0} RHS = (1 - \eta)$ and $\lim_{\gamma \to 1} RHS = +\infty$. Since $0 \le \mu^* < \mu \le 1$, it follows that there exists a *unique* solution for γ verifying $0 \le \gamma \le 1$. Rearranging (A.4) yields the following quadratic equation

$$[(1-\eta)\mu^* - (1-2\eta) - \mu\eta]\gamma^2 + [1-2\eta + 2\mu\eta]\gamma - \mu\eta = 0.$$

Simple computations show that the solution is given by

$$\gamma = \frac{-(1 - 2\eta + 2\mu\eta) + \sqrt{(1 - 2\eta)^2 + 4\mu\eta(1 - \eta)\mu^*}}{2[(1 - \eta)\mu^* - (1 - 2\eta) - \mu\eta]},$$
(A.5)

where it can easily be verified that $0 < 1/2 \le \eta < \gamma \le 1$ for any population distribution verifying $\eta \in [1/2, 1]$.

To compute Q, using $Q = Q^*$ in (A.3) and re-arranging terms, we obtain (20). Finally, plugging (20) in (24), we obtain:

$$g = \frac{\nu \delta \eta [\gamma + \mu (1 - \gamma)] - \gamma \rho}{\gamma (1 + \nu)}.$$

The above expression clearly shows that $\nu > \frac{\rho}{\delta \eta}$ is a necessary and sufficient condition for g > 0 (cf. Assumption 1), since the RHS is strictly increasing in γ .

9.2 Gross-profit and business-stealing effects

By use of equation (3) and the equilibrium value $L_{q_j,t}^X = \frac{L^X}{\gamma QN} = \frac{\eta}{\gamma QN} \left(1 - \frac{1}{Q}\right)$, the business stealing effect $(\nu(1 - X_{q_j,t}/X_{j,t}))$ and the gross profit effect $(\frac{w_t L_{q_j,t}^A}{X_{q_j,t}/X_{j,t}})$ are given by, respectively: $\nu\left(\frac{Q-1}{Q}\right)$ and $w_t \frac{\eta}{\gamma QN}$. Recall that the gross profit effect is increasing in firms' market power (decreasing in the number of firms) while the reverse applies to the business stealing effect. In equilibrium the product of the two effects, hereafter denoted by Γ , is given by (for any N > 2)

$$\Gamma = \nu \left(\frac{Q-1}{Q}\right) w_t \frac{\eta}{\gamma QN} .$$

Normalising $\tau_D = \tau_D^* = \tau_I = 1$, the equilibrium wage is $w = \frac{\gamma}{1-\gamma} \frac{1-\eta}{\eta} > 1$, which is independent of Q. Differentiating the above with respect to Q, we obtain

$$\frac{d\Gamma}{dQ} = -w \frac{\nu \eta}{\gamma N} \left(\frac{Q-1}{Q^3} \right) < 0,$$

implying that the gross profit effect dominates the business stealing effect.

9.3 Comparative statics

9.3.1 Change in γ with respect to μ , μ^* and η

Applying the implicit function theorem to (A.4), we obtain

$$\begin{split} \frac{d\gamma}{d\mu} &= \frac{\frac{(1-\gamma)}{\gamma} \frac{\eta}{(1-\eta)}}{\frac{\mu}{\gamma^2} \frac{\eta}{(1-\eta)} + \frac{\mu^*}{(1-\gamma)^2}} > 0, \\ \frac{d\gamma}{d\mu^*} &= \frac{-\frac{\gamma}{(1-\gamma)}}{\frac{\mu}{\gamma^2} \frac{\eta}{(1-\eta)} + \frac{\mu^*}{(1-\gamma)^2}} < 0, \\ \frac{d\gamma}{d\eta} &= \frac{\left[1 + \mu \frac{(1-\gamma)}{\gamma} + 1 + \mu^* \frac{\gamma}{(1-\gamma)}\right] \frac{1}{(1-\eta)}}{\frac{\mu}{\gamma^2} \frac{\eta}{(1-\eta)} + \frac{\mu^*}{(1-\gamma)^2}} > 0. \end{split}$$

9.3.2 Change in Q with respect to μ , μ^* and η

Note that a change in Q with respect to any parameter x is given by $\frac{dQ}{dx} = \frac{\partial Q}{\partial x} + \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{dx}$ From (20) it readily follows that $\frac{\partial Q}{\partial \gamma} = \frac{\eta \delta(\nu+1)\rho\mu}{[\nu\eta\delta[\gamma+\mu(1-\gamma)]-\gamma\rho]^2} > 0$ and $\frac{\partial Q}{\partial \mu} = \frac{-(1+\nu)\delta\eta\rho(1-\gamma)\gamma}{\{\delta\nu[\gamma+\mu(1-\gamma)]\eta-\rho\gamma\}^2} < 0$. Accordingly, after replacing $\frac{d\gamma}{d\mu}$ and rearranging terms, we obtain,

$$\frac{dQ}{d\mu} = \frac{\partial Q}{\partial \mu} + \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{d\mu} = \frac{\delta \eta \rho (1 - \gamma) \gamma (1 + \nu)}{\{\delta \nu [\gamma + \mu (1 - \gamma)] \eta - \rho \gamma\}^2} \left\{ \frac{\mu \frac{\eta}{(1 - \eta)}}{\mu \frac{\eta}{(1 - \eta)} + \frac{\gamma^2 \mu^*}{(1 - \gamma)^2}} - 1 \right\} < 0.$$

Similarly,

$$\frac{dQ}{d\mu^*} = \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{d\mu^*} < 0.$$

To compute $dQ/d\eta$, let us recall that $L^A = \eta/Q$ (see (18)). Thus, we have

$$\frac{dQ}{d\eta} = \frac{1 - \frac{\eta}{L^A} \frac{dL^A}{d\eta}}{L^A}.$$

Therefore, the sign of $dQ/d\eta$ is the same as the sign of $1 - \frac{\eta}{L^A} \frac{dL^A}{d\eta}$. Using (20), we obtain

$$\frac{dL^A}{d\eta}\frac{\eta}{L^A} = \eta \frac{\left[\nu\delta - \left(\frac{\gamma\rho\mu(1-\gamma)^2}{\gamma^2(1-\eta)\mu^* + \eta\mu(1-\gamma)^2}\right)\frac{1}{[\gamma+\mu(1-\gamma)](1-\eta)}\right]}{\left[\nu\eta\delta - \frac{\gamma\rho}{[\gamma+\mu(1-\gamma)]}\right]}.$$

Simplifying the above expression, $1 - \frac{\eta}{L^A} \frac{dL^A}{d\eta} < 0$ is equivalent to the following condition

$$\frac{\mu(1-\gamma)}{\gamma}\frac{\eta}{(1-\eta)^2}<\frac{\gamma}{(1-\gamma)}\mu^*+\frac{\eta\mu(1-\gamma)}{(1-\eta)\gamma}.$$

Using the equilibrium condition (19), substitute $\mu^* \frac{\gamma}{1-\gamma} = \frac{\eta}{1-\eta} \left[1 + \mu \frac{(1-\gamma)}{\gamma} \right] - 1$ in the equation above to obtain, after simple manipulations

$$\frac{\mu(1-\gamma)}{\gamma}\frac{\eta}{(1-\eta)} < 1,$$

which is always satisfied since $0 < \mu < 1$ and $\gamma > \eta$. Therefore, $dQ/d\eta < 0$.

9.3.3 Change in L_A with respect to η

In a similar way as before, we can use $L^A = \eta/Q$ (see (18)) and (20) to obtain

$$\frac{dL^A}{d\eta} = \frac{1}{\delta(\nu+1)} \left[\nu \delta - \frac{\rho}{\left[\gamma + \mu(1-\gamma)\right]} \frac{\mu}{\left[\gamma + \mu(1-\gamma)\right]} \frac{d\gamma}{d\eta} \right].$$

After substituting for $d\gamma/d\eta > 0$, we obtain

$$\frac{dL^{4}}{d\eta} = \frac{1}{\delta(\nu+1)} \left[\nu\delta - \left(\frac{\gamma\rho\mu(1-\gamma)^{2}}{\gamma^{2}(1-\eta)\mu^{*} + \eta\mu(1-\gamma)^{2}} \right) \frac{1}{[\gamma+\mu(1-\gamma)](1-\eta)} \right].$$

From (20), we know that $\nu\delta > \frac{\gamma\rho}{\eta[\gamma+\mu(1-\gamma)]}$. Therefore, let us show

$$\left(\frac{\gamma\rho\mu(1-\gamma)^2}{\gamma^2(1-\eta)\mu^*+\eta\mu(1-\gamma)^2}\right)\frac{1}{[\gamma+\mu(1-\gamma)](1-\eta)}<\frac{\gamma\rho}{\eta[\gamma+\mu(1-\gamma)]},$$

implying that $dL^4/d\eta > 0$. Simplifying the previous expression, we obtain the following condition

$$\frac{\eta^2 \mu (1 - \gamma)}{(1 - \eta)^2 \gamma} < \frac{\gamma}{(1 - \gamma)} \mu^*$$

Using the equilibrium condition (19), substitute $\mu^* \frac{\gamma}{1-\gamma} = \frac{\eta}{1-\eta} \left[1 + \mu \frac{(1-\gamma)}{\gamma} \right] - 1$ into the equation above to obtain the following inequality

$$\frac{\mu(1-\gamma)}{\gamma}\frac{\eta}{(1-\eta)} < 1,$$

which is always satisfied as $\gamma > \eta$ and $0 < \mu < 1$. Therefore, $dL^A/d\eta > 0$.

9.3.4 Change in g with respect to μ and μ^*

Recall that the equilibrium growth rate is given by

$$g = \frac{\delta\nu[\gamma + \mu(1-\gamma)]\frac{\eta}{\gamma} - \rho}{(1+\nu)}.$$

Differentiating with respect to μ and μ^* yields, respectively

$$\frac{dg}{d\mu} = \frac{\nu}{1+\nu} \left\{ \delta \eta \frac{\mu^*}{\mu^* \frac{\gamma}{1-\gamma} + \mu \frac{\eta}{1-\eta} \frac{1-\gamma}{\gamma}} \right\} > 0, \\ \frac{dg}{d\mu^*} = \frac{\nu}{1+\nu} \left\{ \delta \eta \frac{\mu}{\mu^* \frac{\gamma}{1-\gamma} + \mu \frac{\eta}{1-\eta} \frac{1-\gamma}{\gamma}} \right\} > 0.$$

9.4 Wage

9.4.1 Proof that $w > w^* = 1$

Equation (21) readily shows that its LHS is strictly increasing with $\lim_{w\to 0} LHS = \eta \tau_I/\tau_D^* < \lim_{w\to 0} RHS = +\infty$ and $\lim_{w\to +\infty} LHS = +\infty > \lim_{w\to +\infty} RHS = \gamma \eta \tau_I/[(1-\gamma)(1-\eta)\tau_D]$. Therefore, the solution for the level of wage is unique. Moreover, under the assumption $\tau_I/\tau_D > \tau_I/\tau_D^*$ and the property $\gamma > \eta \ge 1/2$, we can check

$$\lim_{w \to 1} \mathit{LHS} = \frac{\eta}{(1-\eta)} \left[\frac{\eta}{(1-\eta)} + \frac{\tau_I}{\tau_D^*} \right] \ < \ \lim_{w \to 1} \mathit{RHS} = \frac{\gamma}{1-\gamma} \left[\frac{\eta}{(1-\eta)} \frac{\tau_I}{\tau_D} + 1 \right] \ .$$

i.e., the intersection between LHS and RHS in (21) necessarily occurs at w > 1.

9.4.2 Proof that $dw/d\eta > 0$

Manipulating (21), we obtain:

$$\frac{\eta}{1-\eta}w^2 + \left(\frac{\tau_I}{\tau_D^*} - \frac{\gamma}{1-\gamma}\frac{\tau_I}{\tau_D}\right)w - \left(\frac{1-\eta}{\eta}\right)\frac{\gamma}{1-\gamma} = 0. \tag{A.6}$$

Thus, the solution for the wage is given by

$$w = \frac{\frac{\gamma}{1-\gamma} \frac{\tau_I}{\tau_D} - \frac{\tau_I}{\tau_D^*} + \left[\left(\frac{\gamma}{1-\gamma} \frac{\tau_I}{\tau_D} - \frac{\tau_I}{\tau_D^*} \right)^2 + \frac{4\gamma}{1-\gamma} \right]^{1/2}}{2 \frac{\eta}{1-\eta}} . \tag{A.7}$$

Applying the implicit function theorem to (A.6), and substituting w by its value given by (A.7) in the denominator, we obtain

$$\frac{dw}{d\eta} = -\frac{\frac{\partial F}{\partial \eta}}{\left[\left(\frac{\gamma}{1-\gamma}\frac{\tau_I}{\tau_D} - \frac{\tau_I}{\tau_D^*}\right)^2 + \frac{4\gamma}{1-\gamma}\right]^{1/2}},$$

where

$$\frac{\partial F}{\partial \eta} = -\frac{w}{(1-\gamma)^2} \frac{\tau_I}{\tau_D} \left[\frac{d\gamma}{d\eta} - \frac{(1-\gamma)^2}{(1-\eta)^2} \frac{\tau_D}{\tau_I} w \right] - \frac{(1-\eta)}{\eta} \frac{1}{(1-\gamma)^2} \left[\frac{d\gamma}{d\eta} - \frac{1-\gamma}{\eta} \frac{\gamma}{1-\eta} \right].$$

Note that, if $\frac{d\gamma}{d\eta} > \frac{1-\gamma}{\eta} \frac{\gamma}{1-\eta}$ and $\frac{(1-\gamma)^2}{(1-\eta)^2} \frac{\tau_D}{\tau_I} w < \frac{1-\gamma}{\eta} \frac{\gamma}{1-\eta}$, then $\frac{\partial F}{\partial \eta}$ would be negative. Starting from the latter, where we substitute w by its value given by (A.7), after some

Starting from the latter, where we substitute w by its value given by (A.7), after some manipulations, we obtain the following inequality

$$-4\frac{\gamma}{1-\gamma}\frac{\tau_I}{\tau_D}\frac{\tau_I}{\tau_D^*} + \frac{4\gamma}{1-\gamma} < 0,$$

which is always satisfied as $\tau_I > \tau_D^* > \tau_D$. Next, let us show that $\frac{d\gamma}{d\eta} > \frac{1-\gamma}{\eta} \frac{\gamma}{1-\eta}$. Noting that we can write

$$\frac{d\gamma}{d\eta} = \frac{\left[1 + \mu \frac{(1-\gamma)}{\gamma} + 1 + \mu^* \frac{\gamma}{(1-\gamma)}\right] \frac{1}{(1-\eta)}}{\frac{\mu}{\gamma^2} \frac{\eta}{(1-\eta)} + \frac{\mu^*}{(1-\gamma)^2}},$$

after some computations, we obtain

$$\frac{d\gamma}{d\eta} > \frac{1-\gamma}{\eta} \frac{\gamma}{1-\eta} \Longleftrightarrow \frac{\eta}{[(1-\eta)\mu^* + \eta]} > \gamma.$$

To show that the above condition is always verified, let us recall that (A.4) is given by

$$(1-\eta)\left[1+\mu^*\frac{\gamma}{(1-\gamma)}\right]=\eta\left[1+\mu\frac{(1-\gamma)}{\gamma}\right].$$

Now let us define

$$LHS(\gamma) = (1 - \eta) \left[1 + \mu^* \frac{\gamma}{(1 - \gamma)} \right];$$
$$RHS(\gamma) = \eta \left[1 + \mu \frac{(1 - \gamma)}{\gamma} \right].$$

We know that $LHS(\gamma)$ is strictly increasing and $RHS(\gamma)$ is strictly decreasing. Moreover, we can easily check that

$$\mathit{RHS}\left(\frac{\eta}{\mu^*(1-\eta)+\eta}\right) \, < \, \mathit{LHS}\left(\frac{\eta}{\mu^*(1-\eta)+\eta}\right).$$

As in equilibrium we must have $RHS(\gamma) = LHS(\gamma)$, this implies that $\gamma < \eta/[(1-\eta)\mu^* +$ η]. That is, the intersection between $RHS(\gamma)$ and $LHS(\gamma)$ occurs for a value of γ such that: $\gamma < \frac{\eta}{[(1-\eta)\mu^*+\eta]}$. Therefore, $dw/d\eta > 0$.

9.4.3 Evaluating $\lim_{\eta\to 0} w$ and $\lim_{\eta\to 1} w$

In this section, we compute the level of wage in the limit cases where η tends to 0 or 1. Using (A.5) we can obtain

$$\lim_{\eta \to 0} \frac{d\gamma}{d\eta} = \mu < 1,$$

and

$$\lim_{\eta \to 1} \frac{d\gamma}{d\eta} = \mu^* < 1.$$

Then, using the above results, along with l'Hôpital's rule applied to (A.7), we obtain

$$\lim_{\eta \to 0} w = \lim_{\eta \to 0} \frac{d\gamma}{d\eta} \left(\frac{\tau_I}{\tau_D^*}\right)^{-1} = \mu \frac{\tau_D^*}{\tau_I} < 1 \,,$$

and

$$\lim_{\eta \to 1} w = \left(\frac{d\gamma}{d\eta}\right)^{-1} \frac{\tau_I}{\tau_D} = \frac{1}{\mu^*} \frac{\tau_I}{\tau_D} > 1.$$

9.4.4 Change in w with respect to μ , μ^* , $\tau_{\rm I}$, $\tau_{\rm D}$, $\tau_{\rm D}^*$

Recall that the wage is implicitly given by (see (21))

$$\frac{\eta}{(1-\eta)} \left[\frac{\eta}{(1-\eta)} w + \frac{\tau_I}{\tau_D^*} \right] = \frac{\gamma}{1-\gamma} \left[\frac{\eta}{(1-\eta)} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right].$$

Applying the implicit function theorem to the above, we obtain

$$\begin{split} \frac{dw}{d\mu} &= \frac{\frac{1}{(1-\gamma)^2} \left[\frac{\eta}{(1-\eta)} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right] \frac{d\gamma}{d\mu}}{\frac{\eta^2}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} > 0 \,, \\ \frac{dw}{d\mu^*} &= \frac{\frac{1}{(1-\gamma)^2} \left[\frac{\eta}{(1-\eta)^2} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right] \frac{d\gamma}{d\mu^*}}{\frac{\eta^2}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} < 0 \,, \\ \frac{dw}{d\tau_I} &= -\frac{\frac{\eta}{(1-\eta)} \left(\frac{1}{\tau_D^*} - \frac{\gamma}{1-\gamma} \frac{1}{\tau_D} \right)}{\frac{\eta^2}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} > 0 \,, \\ \frac{dw}{d\tau_D} &= -\frac{\frac{\gamma}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}}{\frac{1-\gamma}{(1-\eta)} \frac{\tau_I}{(\tau_D)^2}} < 0 \,, \\ \frac{dw}{d\tau_D^*} &= \frac{\frac{\eta}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}}{\frac{\tau_I}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} > 0 \,. \end{split}$$

9.5 Proof of Proposition 4

Using (26), define the following function,

$$F(\underline{\eta}) = \frac{1}{\rho} \left[\log w + \gamma \log \left(\frac{\tau_I}{\tau_D} \right) + (1 - \gamma) \log \left(\frac{\tau_D^*}{\tau_I} \right) \right] - m = 0.$$

Since μ is set higher than μ^* by default (and it needs to be fulfilled at all times), and both parameters are bounded from above, to study the effect of changes in spillovers we simply look at a decrease in μ^* for a given μ . This is as if spillovers are (in relative terms) more potent in the North. Thus, applying the implicit function theorem to $F(\bullet)$ above, we obtain,

$$\frac{d\underline{\eta}}{d\mu^*} = -\frac{\frac{1}{w}\frac{\partial w}{\partial \mu^*} + \frac{\partial \gamma}{\partial \mu^*}\log\frac{\tau_L}{\tau_D}\frac{\tau_L}{\tau_D^*}}{\frac{1}{w}\frac{\partial w}{\partial \underline{\eta}} + \frac{\partial \gamma}{\partial \underline{\eta}}\log\left(\frac{\tau_L}{\tau_D}\frac{\tau_L}{\tau_D^*}\right)} > 0.$$

The denominator of $\frac{d\underline{\eta}}{d\mu^*}$ is positive while the numerator is negative and, overall, the effect is positive.

9.6 Proof of Proposition 6

Apply the implicit function theorem to (21) to obtain $\frac{dw}{d\gamma} = \frac{\frac{1}{(1-\gamma)^2} \left[\frac{\eta}{(1-\eta)^2 D^2 D^2} + \frac{\eta}{w}\right]}{\left(\frac{\eta}{1-\eta}\right)^2 + \frac{\gamma}{1-\gamma_w^2}} > 0$. Since $\frac{d\gamma}{d\mu} > 0$ and $\frac{d\gamma}{d\mu^*} < 0$, then w is positively correlated with μ and negatively correlated with μ^* . Furthermore, since $\frac{d\gamma}{d\eta} > 0$ it follows that w is also positively correlated with η .

9.7 Welfare

Using (6), (14), E = w and $E^* = w^* = 1$, in steady state, the individual lifetime utility in the North reads

$$\rho U = \log \left[w^{1-\gamma} \frac{\left(A_{q_j,0} \right)^{\nu} (Q-1)}{NQ} \left(\frac{1}{\tau_D} \right)^{\gamma} \left(\frac{1}{\tau_I} \right)^{1-\gamma} \right] + \nu \frac{g}{\rho} ,$$

Similarly, for the South, we have

$$\rho U^* = \log \left[w^{-\gamma} \frac{\left(A_{q_j,0} \right)^{\nu} (Q-1)}{NQ} \left(\frac{1}{\tau_I} \right)^{\gamma} \left(\frac{1}{\tau_D^*} \right)^{1-\gamma} \right] + \nu \frac{g}{\rho} .$$

Differentiating the above expressions with respect to μ^* yields (30) and (31) in the main text.

9.7.1 Proof that $\frac{dU^*}{d\mu^*} > 0$

$$\frac{dU^*}{d\mu^*} = \frac{1}{\rho} \left\{ \frac{1/Q^2}{1 - 1/Q} \frac{dQ}{d\mu^*} + \frac{1 - \gamma}{w} \frac{dw}{d\mu^*} - \frac{d\gamma}{d\mu^*} \log w + \frac{d\gamma}{d\mu^*} \left(\log \frac{1}{\tau_I} - \log \frac{1}{\tau_D^*} \right) + \frac{\nu}{\rho} \frac{dg}{d\mu^*} \right\},$$

where $1 < au_D^* < au_I, Q = \frac{(1+\nu)[\gamma + \mu(1-\gamma)]\frac{\delta \eta}{\gamma}}{\nu[\gamma + \mu(1-\gamma)]\frac{\delta \eta}{\gamma} - \rho} > 1, \frac{dQ}{d\mu^*} = \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{d\mu^*} < 0, \frac{dw}{d\mu^*} = \frac{\frac{1}{(1-\gamma)^2} \left[\frac{\eta}{(1-\eta)\Gamma} + \frac{1}{w}\right]\frac{d\gamma}{d\mu^*}}{\frac{\eta^2}{(1-\eta)^2} + \frac{1}{1-\gamma w^2}} < \frac{1}{(1-\eta)^2} \left[\frac{\eta}{(1-\eta)^2} + \frac{1}{1-\gamma w^2}\right]^{\frac{1}{2}} = \frac{1}{(1-\eta)^2} \left[\frac{\eta}{(1-\eta)^2} + \frac{1}{1-\eta} + \frac{1}{1-$

 $0, \frac{d\gamma}{d\mu^*} < 0$ and $\frac{dg}{d\mu^*} = -\frac{g}{Q}\frac{dQ}{d\mu^*} - \delta\eta \frac{1}{Q}\frac{\mu}{\gamma^2}\frac{d\gamma}{d\mu^*} > 0$. Therefore, for $\frac{dU^*}{d\mu^*} > 0$ to hold it suffices to show that $\frac{1/Q^2}{1-1/Q}\frac{dQ}{d\mu^*} - \frac{g}{Q}\frac{dQ}{d\mu^*} > 0$, that is the dynamic competition effect outweighs the static competition effect. Since $\frac{dQ}{d\mu^*} < 0$, we need to show that

$$\frac{1}{Q-1} < \frac{\nu}{\rho}g.$$

Substituting for $\frac{1}{Q-1} = \frac{\nu[\gamma + \mu(1-\gamma)]\delta\eta - \gamma\rho}{[\gamma + \mu(1-\gamma)]\delta\eta + \gamma\rho}$ and $g = \frac{\nu[\gamma + \mu(1-\gamma)]\frac{\delta\eta}{\gamma} - \rho}{(1+\nu)}$, we obtain

$$\nu > \left(\frac{\rho(1+\nu)\gamma}{\nu[\gamma+\mu(1-\gamma)]\delta\eta-\rho\gamma}\right) \left(\frac{\nu[\gamma+\mu(1-\gamma)]\delta\eta-\gamma\rho}{[\gamma+\mu(1-\gamma)]\delta\eta+\gamma\rho}\right).$$

Simplifying yields,

$$\nu > \frac{\gamma \rho}{[\gamma + \mu(1 - \gamma)]\delta \eta}$$
.

This is always satisfied under assumption 1.

9.7.2 Proof that $\frac{dU}{du^*} > 0$

Assuming negligible transport costs ($\tau_D = \tau_D^* = \tau_I \to 1$) we have

$$\frac{dU}{d\mu^*} = \frac{1}{\rho} \left[\frac{1/Q^2}{1-1/Q} \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{d\mu^*} + \frac{1-\gamma}{w} \frac{dw}{d\mu^*} - \frac{d\gamma}{d\mu^*} \log w + \frac{\nu}{\rho} \frac{d\mathbf{g}}{d\mu^*} \right],$$

where $\frac{dg}{d\mu^*} = -\frac{g}{Q}\frac{dQ}{d\mu^*} - \delta\eta \frac{1}{Q}\frac{\mu}{\gamma^2}\frac{d\gamma}{d\mu^*}$. Under assumption 1 (cf. proof above) we know that $\frac{1/Q^2}{1-1/Q}\frac{dQ}{d\mu^*} - \frac{g}{Q}\frac{dQ}{d\mu^*} > 0$. Hence for $\frac{dU}{d\mu^*} > 0$ to hold it suffices to show that $\frac{1-\gamma}{w}\frac{dw}{d\mu^*} \leq \log w \frac{d\gamma}{d\mu^*} + \frac{\nu}{\rho}\delta\eta \frac{1}{Q}\frac{\mu}{\gamma^2}\frac{d\gamma}{d\mu^*}$, that is the gains associated with the re-location of sectors outweighs the losses associated with the decrease in the relative wage.

Under negligible transport costs the wage equation (21) $implies w = \frac{\left(\frac{2\gamma-1}{1-\gamma}\right) + \left[\frac{4\gamma^2+1-4\gamma}{(1-\gamma)^2} + \frac{4\gamma}{1-\gamma}\right]^{1/2}}{2\frac{\eta}{1-\eta}} = \frac{\gamma}{1-\gamma} \frac{1-\eta}{\eta} > 1$ and $\frac{dw}{d\mu^*} = \frac{1}{(1-\gamma)^2} \frac{1-\eta}{\eta} \frac{d\gamma}{d\mu^*}$, hence we need to show that

$$\frac{1}{\gamma} - 1 + \frac{1 - \gamma}{\gamma} \frac{\eta}{1 - \eta} - \frac{\nu}{\rho} \delta \eta \frac{1}{Q \gamma^2} \le 0,$$

which simplifies to

$$\frac{1-\gamma}{1-\eta} \le \frac{\nu}{\rho} \delta \eta \frac{1}{Q} \frac{\mu}{\gamma}.$$

Replacing Q and simplifying

$$\frac{c}{d}\frac{1-\gamma}{1-\eta}\frac{(1+\nu)[\gamma+\mu(1-\gamma)]}{\nu[\gamma+\mu(1-\gamma)]\frac{\delta\eta}{\gamma}-\rho}\mu\Longleftrightarrow\frac{\gamma\rho(1-\gamma)(1+\nu)[\gamma+\mu(1-\gamma)]}{\nu(1-\eta)\mu\{\nu[\gamma+\mu(1-\gamma)]\delta\eta-\gamma\rho\}}\leq\,1.$$

Denote by $\hat{\nu}$ the solution of the implicit function $\frac{\gamma\rho(1-\gamma)(1+\hat{\nu})[\gamma+\mu(1-\gamma)]}{\hat{\nu}(1-\eta)\mu\{\hat{\nu}[\gamma+\mu(1-\gamma)]\delta\eta-\gamma\rho\}}=1$; then $\frac{dU}{du^*}>0$ for any $\nu\leq\hat{\nu}$.

9.8 Asymmetries in productivity

Here we postulate $\delta > \delta^*$ (i.e., R&D productivity in the North is greater than in the South) and $\mu = \mu^* < 1$. In this case, the condition equalising growth rates $(g = g^*)$ reads as

$$\delta \eta \left[1 + \frac{\mu(1-\gamma)}{\gamma} \right] = \delta^*(1-\eta) \left[1 + \frac{\gamma \mu}{(1-\gamma)} \right].$$

Accordingly, the location decision of sectors depends on R&D productivity.

It can be easily checked that the left hand side (hereafter LHS) is strictly decreasing, with $\lim_{\gamma \to 0} LHS = +\infty$ and $\lim_{\gamma \to 1} LHS = \delta \eta$. Similarly, the right hand side (hereafter, RHS) is strictly increasing, with $\lim_{\gamma \to 0} RHS = \delta^*(1-\eta)$ and $\lim_{\gamma \to 1} RHS = +\infty$. Under the assumption $0 \le \delta \eta \le 1$, it follows that there exists a unique solution for γ verifying $0 \le \gamma \le 1$. Notice, also, that $\delta \eta < 1$ is not stringent as $\eta \le 1$ and we can expect δ to be much lower than 1.

Note that when $\eta = 1/2$, the equation above can be written as

$$\delta^* \mu \left(\frac{\gamma}{1-\gamma}\right)^2 - (\delta - \delta^*) \left(\frac{\gamma}{1-\gamma}\right) - \delta \mu = 0.$$

Solving for $\frac{\gamma}{1-\gamma}$ we obtain

$$\frac{\gamma}{1-\gamma} = \frac{(\delta-\delta^*) + \sqrt{(\delta-\delta^*)^2 + 4\delta^*\mu(\delta-\delta^*)}}{2\delta^*\mu}.$$

Therefore when $\eta = 1/2, \gamma = X$ where

$$X = \frac{\left(\delta - \delta^*\right) + \sqrt{\left(\delta - \delta^*\right)^2 + 4\delta^*\mu(\delta - \delta^*)}}{2\delta^*\mu + \left(\delta - \delta^*\right) + \sqrt{\left(\delta - \delta^*\right)^2 + 4\delta^*\mu(\delta - \delta^*)}}.$$

Furthermore, X > 1/2 if, after some simplifications, $(\delta - \delta^*) + \sqrt{(\delta - \delta^*)^2 + 4\delta^* \mu(\delta - \delta^*)} > 2\delta^* \mu$.

Since $\delta > \delta^*$, we can as well show that $(\delta - \delta^*) + \sqrt{(\delta - \delta^*)^2 + 4\delta^*\mu(\delta - \delta^*)} > \delta^*\mu + \delta\mu$. Simplifying, the latter condition becomes $\delta(1-\mu) - \delta^*(1-\mu) + \sqrt{(\delta - \delta^*)^2 + 4\delta^*\mu(\delta - \delta^*)} > 0$, which is always satisfied. This proves that $\gamma > 1/2$ when $\eta = 1/2$ as in the case analysed in the main section of the paper.

Finally, by applying the implicit function theorem, it can be easily checked that $\frac{d\gamma}{d\delta^*} < 0$. Therefore, the properties of the model described for μ^* apply to δ^* .

9.9 Intra-regional equilibrium under heterogeneity and population growth

The problem of individuals remains the same as in the main text. For firms, the current value Hamiltonian becomes

$$CVH_{q_{j},t} = \left\{ X_{q_{j},t} \left[\Omega_{t} \left(\sum_{q_{j}=1}^{Q_{j,t}} X_{q_{j},t} \right)^{-1} - w_{t} (A_{q_{j},t})^{-\nu} \right] - w_{t} L_{q_{j},t}^{A} \right\}.$$

$$+ \xi_{t} \delta L_{q_{i},t}^{A} K_{t}$$

The first order conditions are given by: $\partial CVH_{q_j,t}/\partial X_{q_j,t}=0$, $\partial CVH_{q_j,t}/\partial L_{q_j,t}^A=0$ and $\partial CVH_{q_j,t}/\partial A_{q_j,t}=-\xi_t^{\bullet}+r_t\xi_t$. The transversality condition is $\lim_{t\to\infty}\xi_tA_{q_j,t}e^{-\int_0^tr_udu}=0$. After simple manipulations, the price of variety j is given by

$$p_{j,t} = \frac{w_t (A_{q_j,t})^{-\nu}}{(1 - X_{q_j,t}/X_{i,t})}.$$
 (A.8)

The zero profit condition reads as

$$\frac{w_t L_{q_j,t}^X}{(1 - X_{q_j,t}/X_{j,t})} - w_t L_{q_j,t}^X - w_t L_{q_j,t}^A \le 0.$$
(A.9)

The rate of return to R&D is now given by

$$r_t = \delta \nu L_{q_j,t}^X \frac{\tilde{A}_t}{(A_{q_j,t})} + \frac{\xi_t^{\bullet}}{\xi_t}.$$
 (A.10)

Equations giving the relation between the amounts of labour allocated to the production of varieties, L_t^X , to R&D, L_t^A , and the number of firms in each sector, Q, are respectively

$$\frac{QL_t^X}{(Q-1)} - L_t^X - L_t^A = 0,$$

$$L_t^X = \eta L_t^w \left(1 - \frac{1}{Q}\right),$$

and

$$L_t^A = \eta L_t^w \frac{1}{Q}.$$

Then, we can compute the number of sectors located in the North, γ , the number of firms in each sector, Q, and the steady-state growth rate, g, respectively

$$\eta \left[1 + \mu \frac{(1-\gamma)}{\gamma} \frac{1}{\overline{A}_t/\overline{A}_t^*} \right] = (1-\eta) \left[1 + \mu^* \frac{\gamma}{(1-\gamma)} \overline{A}_t/\overline{A}_t^* \right], \tag{A.11}$$

$$Q = \frac{(1+\nu)\delta \frac{\eta L_t^w}{\gamma N_t} \left[\gamma + \mu (1-\gamma) \frac{1}{\overline{A_t/A_t^*}} \right]}{\delta \nu \frac{\eta L_t^w}{\gamma N_t} \left[\gamma + \mu (1-\gamma) \frac{1}{\overline{A_t/A_t^*}} \right] - (\rho - \omega)},$$
(A.12)

$$g = \frac{1}{1+\nu} \left\{ \nu \frac{\eta \delta L_t^w}{\gamma N_t} \left[\gamma + \mu (1-\gamma) \frac{1}{\overline{A}_t / \overline{A}_t^*} \right] - (\rho - \omega) \right\}. \tag{A.13}$$

Finally, the wage in the North and South verify

$$w_t = \frac{\gamma}{L_t^w} \left[\frac{E_t}{\tau_D} + \frac{(1 - \eta)}{\eta} \frac{E^*}{\tau_I} \right],$$

and

$$w_t^* = \frac{(1-\gamma)}{L_t^w} \left[\frac{\eta}{(1-\eta)} \frac{E}{\tau_I} + \frac{E^*}{\tau_D^*} \right],$$

implying that w is solution of the following equation

$$\frac{\eta}{(1-\eta)} \left[\frac{\eta}{(1-\eta)} w + \frac{\tau_I}{\tau_D^*} \right] = \frac{\gamma}{1-\gamma} \left[\frac{\eta}{(1-\eta)} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right]. \tag{A.14}$$

It is worth emphasising here that the equation is exactly the same as in the main text, although we should keep in mind that γ now depends on the relative average amount of knowledge between regions, $\overline{A}_t/\overline{A}_t^* \equiv \Theta$. Equally, the comparative statics identified in section 9.3 retain the same signs, as shown below.

9.9.1 Comparative statics

Applying the implicit function theorem to (A.11), we obtain

$$\begin{split} \frac{d\gamma}{d\mu} &= \frac{\frac{(1-\gamma)}{\gamma} \frac{\eta}{(1-\eta)} \frac{1}{\Theta}}{\frac{\mu}{\gamma^2} \frac{\eta}{(1-\eta)} \frac{1}{\Theta} + \frac{\mu^*}{(1-\gamma)^2} \Theta} > 0, \\ &\frac{d\gamma}{d\mu^*} = -\frac{\frac{\gamma}{1-\gamma} \Theta}{\frac{\mu}{\gamma^2} \frac{\eta}{(1-\eta)} \frac{1}{\Theta} + \frac{\mu^*}{(1-\gamma)^2} \Theta} < 0, \\ &\frac{d\gamma}{d\Theta} = -\frac{\mu \frac{(1-\gamma)}{\gamma} \frac{\eta}{(1-\eta)} \frac{1}{\Theta^2} + \mu^* \frac{\gamma}{1-\gamma}}{\frac{\mu}{\gamma^2} \frac{\eta}{(1-\eta)} \frac{1}{\Theta} + \frac{\mu^*}{(1-\gamma)^2} \Theta} = -\frac{\gamma(1-\gamma)}{\Theta} < 0. \end{split}$$

Note that $\frac{d\gamma}{d\Theta} < 0$ is consistent with the assumption of decreasing returns to innovation, implying that an increase in average knowledge in the North relative to the South makes the advanced region relative less productive at producing new knowledge and, thereby, a relatively less attractive location for firms. This also implies that, $\gamma > \eta$ might not be satisfied if the difference in average knowledge between regions is large enough. To rule out such a case, we im-

pose the following restriction: $\frac{-\left[1-2\eta+2\eta\mu_{\overline{\Theta}}^{\frac{1}{2}}\right]+\sqrt{\left[1-2\eta+2\mu\eta_{\overline{\Theta}}^{\frac{1}{2}}\right]^{2}+4\mu\eta_{\overline{\Theta}}^{\frac{1}{2}}\left[(1-\eta)\mu^{*}\Theta-(1-2\eta)-\eta\mu_{\overline{\Theta}}^{\frac{1}{2}}\right]}}{2\left[(1-\eta)\mu^{*}\Theta-(1-2\eta)-\eta\mu_{\overline{\Theta}}^{\frac{1}{2}}\right]}>\eta,$ which is fulfilled for Θ not too large.

Differentiating $\frac{d\gamma}{du^*}$ above with respect to Θ , after some manipulations, gives

$$\frac{d^2\gamma}{d\mu^*d\Theta} = \frac{\frac{\gamma}{1-\gamma}\Theta}{\frac{\mu}{\gamma^2}\frac{\eta}{(1-\eta)}\frac{1}{\Theta} + \frac{\mu^*}{(1-\gamma)^2}\Theta} \frac{2\gamma-1}{\gamma(1-\gamma)}\frac{d\gamma}{d\Theta} = \frac{d\gamma}{d\mu^*}\frac{(2\gamma-1)}{\Theta} < 0.$$

Moving on to the effect on the equilibrium number of firms, recall that a change in Q with respect to any parameter x is given by $\frac{dQ}{dx} = \frac{\partial Q}{\partial x} + \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{dx}$ From (A.12) it readily follows that

$$\frac{\partial \underline{Q}}{\partial \gamma} = \frac{(1+\nu)\delta^{\eta L_t^w}_{N_t} \left[\frac{1+\mu}{\gamma} \frac{1}{\Theta}\right](\rho-\omega)}{\left[\delta \nu^{\eta L_t^w}_{N_t} \left[1+\mu^{(1-\gamma)}_{\gamma} \frac{1}{\Theta}\right]-(\rho-\omega)\right]^2} > 0 \quad \text{and} \quad \frac{\partial \underline{Q}}{\partial \mu} = -\frac{(1+\nu)\delta^{\eta L_t^w}_{N_t} \left[1+\mu^{(1-\gamma)}_{\gamma} \frac{1}{\Theta}\right](\rho-\omega)}{\left[\delta \nu^{\eta L_t^w}_{N_t} \left[1+\mu^{(1-\gamma)}_{\gamma} \frac{1}{\Theta}\right]-(\rho-\omega)\right]^2} < 0. \quad \text{Accordingly,}$$

after replacing $\frac{d\gamma}{du}$ and rearranging terms, we obtain

$$\begin{split} \frac{dQ}{d\mu} &= \frac{\partial Q}{\partial \mu} + \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{d\mu} = -\frac{(1+\nu)\delta \frac{\eta L_t^w}{N_t} \frac{(1-\gamma)}{\gamma} (\rho - \omega) \frac{1}{\Theta}}{\left[\delta \nu \frac{\eta L_t^w}{N_t} \left[1 + \mu \frac{(1-\gamma)}{\gamma} \frac{1}{\Theta}\right] - (\rho - \omega)\right]^2} \left[\frac{(1-\eta)\mu^* \frac{\gamma^2}{(1-\gamma)^2} \Theta}{\eta \mu \frac{1}{\Theta} + (1-\eta)\mu^* \frac{\gamma^2}{(1-\gamma)^2} \Theta} \right] < 0 \,, \\ \frac{dQ}{d\mu^*} &= \frac{\partial Q}{\partial \gamma} \frac{d\gamma}{d\mu^*} = \frac{(1+\nu)\delta \frac{\eta L_t^w}{N_t} \mu \frac{1}{\gamma^2} \frac{1}{\Theta} (\rho - \omega)}{\left[\delta \nu \frac{\eta L_t^w}{N_t} \left[1 + \mu \frac{(1-\gamma)}{\gamma} \frac{1}{\Theta}\right] - (\rho - \omega)\right]^2} \frac{d\gamma}{d\mu^*} < 0 \,, \end{split}$$

and

$$\frac{dQ}{d\Theta} = \frac{(1+\nu)\delta\frac{\eta L_t^w}{\gamma N_t} \left(\frac{\mu}{\gamma}\right)\frac{1}{\Theta}(\rho-\omega)}{\left[\delta\nu\frac{\eta L_t^w}{\gamma N_t}\left[\gamma+\mu(1-\gamma)\frac{1}{\Theta}\right]-(\rho-\omega)\right]^2} \left[\frac{\gamma(1-\gamma)}{\Theta} + \frac{d\gamma}{d\Theta}\right] = 0,$$

since $\frac{d\gamma}{d\Theta} = -\frac{\gamma(1-\gamma)}{\Theta}$. Note that, firms' heterogeneity has no impact on the number of firms within each sector. Two effects cancel each other out: The positive direct effect from the change in relative average knowledge between regions (captured by the term in Θ) is offset by the negative indirect effect coming from the relocation of sectors (captured by $\frac{d\gamma}{d\Theta}$). Differentiating $\frac{dQ}{du^c}$ above with respect to Θ , after some manipulations, gives

$$\frac{d^2Q}{d\mu^*d\Theta} = \frac{(1+\nu)\delta\frac{\eta L_v^w}{\gamma N_t} \left(\frac{\mu}{\gamma}\right)\frac{1}{\Theta}(\rho-\omega)}{\left[\delta\nu\frac{\eta L_v^w}{\gamma N_t} \left[\gamma + \mu(1-\gamma)\frac{1}{\Theta}\right] - (\rho-\omega)\right]^2} \left[-\frac{1}{\Theta}\frac{d\gamma}{d\mu^*} - \frac{2}{\gamma}\frac{d\gamma}{d\mu^*}\frac{d\gamma}{d\Theta} + \frac{d^2\gamma}{d\mu^*d\Theta}\right] = 0,$$

where we have used $\frac{d\gamma}{d\Theta} = -\frac{\gamma(1-\gamma)}{\Theta}$ and $\frac{d^2\gamma}{d\mu^*d\Theta} = \frac{d\gamma}{d\mu^*}\frac{(2\gamma-1)}{\Theta}$.\$\$ Similarly, differentiating the growth equation (A.13), we obtain

$$\begin{split} \frac{dg}{d\mu} &= \frac{\nu}{\nu+1} \left[\frac{\delta \eta L_t^w}{N_t} \frac{\mu^*}{\mu^* \frac{\gamma}{1-\gamma} \Theta + \mu \frac{\eta}{1-\eta} \frac{1-\gamma}{\gamma} \frac{1}{\Theta}} \right] > 0, \\ \frac{dg}{d\mu^*} &= -\frac{\nu}{\nu+1} \left[\delta \frac{\eta L_t^w}{N_t} \frac{1}{\Theta} \frac{\mu}{\gamma^2} \frac{d\gamma}{d\mu^*} \right] > 0, \\ \frac{dg}{d\Theta} &= -\frac{\nu}{\nu+1} \delta \frac{\eta L_t^w}{N_t} \frac{1}{\Theta} \frac{\mu}{\gamma^2} \left[\frac{\gamma(1-\gamma)}{\Theta} + \frac{d\gamma}{d\Theta} \right] = 0. \end{split}$$

And, differentiating $\frac{dg}{d\mu^*}$ above with respect to Θ , we obtain:

$$\frac{d^2g}{d\mu^*d\Theta} = -\frac{\nu}{\nu+1} \delta \frac{\eta L_t^w}{\gamma N_t} \left(\frac{\mu}{\gamma}\right) \frac{1}{\Theta} \left[-\frac{1}{\Theta} \frac{d\gamma}{d\mu^*} - \frac{2}{\gamma} \frac{d\gamma}{d\Theta} \frac{d\gamma}{d\mu^*} + \frac{d^2\gamma}{d\mu^*d\Theta} \right] = 0.$$

9.9.2 Effects on ΔU

To assess whether a change in Θ mitigates or reinforces the effect of μ^* on the utility differential, ΔU , we need first to look at the effect on the relative wage between regions. Applying the implicit function theorem to (A.14), we obtain

$$\begin{split} \frac{dw}{d\mu} &= \frac{\frac{1}{(1-\gamma)^2} \left[\frac{\eta}{(1-\eta)} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right] \frac{d\gamma}{d\mu}}{\frac{\eta^2}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} > 0, \\ \frac{dw}{d\mu^*} &= \frac{\frac{1}{(1-\gamma)^2} \left[\frac{\eta}{(1-\eta)} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right] \frac{d\gamma}{d\mu^*}}{\frac{\eta^2}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} < 0, \\ \frac{dw}{d\Theta} &= \frac{\left(\frac{\eta}{1-\eta} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right) \frac{1}{(1-\gamma)^2} \frac{d\gamma}{d\Theta}}{\frac{\eta^2}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} = -\frac{w}{\Theta} \frac{\frac{\gamma}{w(1-\gamma)} \left(\frac{\eta}{1-\eta} \frac{\tau_I}{\tau_D} + \frac{1}{w} \right)}{\frac{\eta^2}{(1-\eta)^2} + \frac{\gamma}{1-\gamma} \frac{1}{w^2}} < 0. \end{split}$$

Differentiating $\frac{dw}{du^*}$ above with respect to Θ , yields:

$$\frac{\left(\frac{\eta}{1-\eta}\frac{\tau_{I}}{\tau_{D}} + \frac{1}{w}\right)\left[\frac{2}{(1-\gamma)^{3}}\left(\frac{\eta}{1-\eta}\right)^{2} + \frac{2\gamma}{(1-\gamma)^{4}w^{2}} - \frac{1}{(1-\gamma)^{4}w^{2}}\right]}{\left[\left(\frac{\eta}{1-\eta}\right)^{2} + \frac{\gamma}{1-\gamma}\frac{1}{w^{2}}\right]^{2}} \\ + \frac{\frac{2\gamma}{(1-\gamma)^{3}w^{3}}\left(\frac{\eta}{1-\eta}\frac{\tau_{I}}{\tau_{D}} + \frac{1}{w}\right) - \frac{1}{(1-\gamma)^{2}w^{2}}\left[\left(\frac{\eta}{1-\eta}\right)^{2} + \frac{\gamma}{1-\gamma}\frac{1}{w^{2}}\right]}{\left[\left(\frac{\eta}{1-\eta}\right)^{2} + \frac{\gamma}{1-\gamma}\frac{1}{w^{2}}\right]^{2}} \\ + \frac{\frac{1}{(1-\gamma)^{2}}\left(\frac{\eta}{1-\eta}\frac{\tau_{I}}{\tau_{D}} + \frac{1}{w}\right)}{\left(\frac{\eta}{1-\eta}\right)^{2} + \frac{\gamma}{1-\gamma}\frac{1}{w^{2}}} \frac{d^{2}\gamma}{d\mu^{*}d\Theta}.$$

Substituting $\frac{d^2\gamma}{d\mu^*d\Theta}$, $\frac{dw}{d\Theta}$, $\frac{d\gamma}{d\Theta}$ and $\frac{d\gamma}{d\mu^*}$ derived earlier, and simplifying, we obtain

$$\frac{d^2w}{d\mu^*d\Theta} = -\frac{1}{\Theta} \frac{\left(\frac{\eta}{1-\eta}\right)^4 + \frac{2\gamma^2}{(1-\gamma)^2w^3} \frac{\eta}{1-\eta} \frac{\tau_I}{\tau_D} + \frac{\gamma^2}{(1-\gamma)^2w^4} \frac{dw}{d\mu^*}}{\left[\left(\frac{\eta}{1-\eta}\right)^2 + \frac{\gamma}{1-\gamma} \frac{1}{w^2}\right]^2} \frac{dw}{d\mu^*} > 0.$$

Differentiating (28) with respect to Θ yields

$$\frac{d^2\Delta U}{d\mu^*d\Theta} = \frac{1}{\rho} \left[-\frac{1}{w^2} \frac{dw}{d\Theta} \frac{dw}{d\mu^*} + \frac{1}{w} \frac{d^2w}{d\mu^*d\Theta} + \frac{d^2\gamma}{d\mu^*d\Theta} \log\left(\frac{\tau_I}{\tau_D} \frac{\tau_I}{\tau_D^*}\right) \right].$$

Substituting $\frac{d^2w}{d\mu^*d\Theta}$, $\frac{dw}{d\Theta}$ and $\frac{dw}{d\mu^*}$ derived earlier, and simplifying, we obtain

$$\frac{d^2\Delta U}{d\mu^*d\Theta} = \frac{1}{\rho} \left\{ -\frac{\frac{1}{\Theta}\frac{1}{w}\left(\frac{\eta}{1-\eta}\right)\frac{dw}{d\mu^*}\left[\frac{\gamma}{(1-\gamma)w^2} - \left(\frac{\eta}{1-\eta}\right)^2\right]\left[\frac{\gamma}{w(1-\gamma)}\frac{\tau_I}{\tau_D} - \frac{\eta}{1-\eta}\right]}{\left[\left(\frac{\eta}{1-\eta}\right)^2 + \frac{\gamma}{1-\gamma}\frac{1}{w^2}\right]^2} + \frac{d^2\gamma}{d\mu^*d\Theta}\log\left(\frac{\tau_I}{\tau_D}\frac{\tau_I}{\tau_D^*}\right)\right\}.$$

We know that $\frac{d^2\gamma}{d\mu^*d\Theta} < 0$ and $\frac{dw}{d\mu^*} < 0$. Moreover, we can easily show that $\frac{\gamma}{(1-\gamma)w^2} - \left(\frac{\eta}{1-\eta}\right)^2 < 0$ and $\frac{\gamma}{w(1-\gamma)}\frac{\tau_I}{\tau_D} - \left(\frac{\eta}{1-\eta}\right) > 0$. First, recall that $\frac{\gamma}{(1-\gamma)} > \frac{\eta}{1-\eta}$ because $\gamma > \eta > 1/2$. Moreover, we have: $\lim_{\tau_D = \tau_{D^*} = \tau_I} w = \frac{\gamma}{1-\gamma}\frac{1-\eta}{\eta} > 1$ and $\frac{dw}{d\tau_{D^*}} > 0$ and $\frac{dw}{d\tau_I} > 0$ implying that $w \ge \frac{\gamma}{1-\gamma}\frac{1-\eta}{\eta}$. Keeping these information in mind, note that $\frac{\gamma}{(1-\gamma)w^2} - \left(\frac{\eta}{1-\eta}\right)^2 < 0$ is equivalent to $\frac{1}{w^2} - \frac{1-\gamma}{\gamma}\left(\frac{\eta}{1-\eta}\right)^2 < 0$ which is always verified. Next, let us show that $\frac{\gamma}{w(1-\gamma)}\frac{\tau_I}{\tau_D} - \frac{\eta}{1-\eta} > 0$. Recall that the wage is implicitly given by (see A.14): $\frac{\eta}{(1-\eta)} + \frac{1}{w}\frac{\tau_I}{\tau_D^*} - \frac{1}{u}\frac{\gamma}{\eta}\frac{\tau_I}{v_D} - \frac{\gamma}{1-\gamma}\frac{1-\eta}{\eta}\frac{1}{w^2} = 0 \Rightarrow \frac{1}{w}\frac{\gamma}{1-\gamma}\frac{\tau_I}{\tau_D} - \frac{\eta}{(1-\eta)} = \frac{1}{w}\frac{\tau_I}{\tau_D^*} - \frac{\gamma}{1-\gamma}\frac{1-\eta}{\eta}\frac{1}{w^2}$. Plugging the latter into the inequality $(\frac{\gamma}{w(1-\gamma)}\frac{\tau_I}{\tau_D} - \frac{\eta}{1-\eta} > 0)$ we obtain $\frac{1}{w}\frac{\tau_I}{\tau_D^*} - \frac{\gamma}{1-\gamma}\frac{1-\eta}{\eta}\frac{1}{w^2} > 0 \iff w\frac{\tau_I}{\tau_D^*} > \frac{\gamma}{1-\gamma}\frac{1-\eta}{\eta}$. Given that the latter is always verified, we can conclude that

$$\frac{d^2\Delta U}{du^*d\Theta} < 0.$$

9.9.3 Demographic shocks

As Peretto (2003), we study the effects of demographic shocks on growth and market structure. I.e. we analyse how variables vary when the population growth rate, ω , varies.

Impact on γ

Inspection of (A.11) implies that any demographic shock (i.e. a change in population growth ω) has no impact on γ . Given that γ depends on η , this result is of course due to the fact that population growth is common to both regions.

Impact on Q

From equation (A.12), we have

$$\frac{dQ}{d\omega} = d\frac{\left\{\frac{(\nu+1)\delta\eta_{N_l}^{L_w^w}\left[1+\mu\frac{(1-\gamma)(\overline{d}_l^*)}{7}\frac{1}{A_t}\right]}{\delta\nu\eta_{N_l}^{L_w^w}\left[1+\mu\frac{(1-\gamma)(\overline{d}_l^*)}{7}\frac{1}{A_t}\right]-(\rho-\omega)}\right\}}{d\omega < 0}.$$

A greater population growth leads to a reduction of firms in each sector, i.e. less competition.

Impact on L_t^A/L_t^w As $L_t^A = \eta L_t^w \frac{1}{Q}$, we obviously have $\frac{dL_t^A/L_t^w}{d\omega} > 0$. As population growth increases, firms allocate a greater share of their workforce to R&D.

Impact on g

From equation (A.13), we obtain $\frac{dg}{d\omega} > 0$. A greater population growth leads to a higher level of growth.

Spatial equilibrium

Note that wage does not depend on population growth: $dw/d\omega = 0$. Therefore, from (A.14), we directly conclude that there is no impact of a demographic shock on the spatial equilibrium.

Acknowledgement

We thank Mike Bleaney, Carl Davidson, Gianni De Fraja, Klaus Desmet, Huw Dixon, Paolo Epifani, Gino Gancia, David Hemous, Chien-Yu Huang, Gianmario Impulliti, Oleg Itskhoki, Vahagn Jerbashian, Omar Licandro, Pietro Peretto, Xavier Raurich, Yibai Yang, Piercarlo Zanchettin, and participants at various conferences and seminars. We are particularly grateful to the editor and one anonymous referee for their comments and inputs. The usual disclaimers apply.

References

- Acs, Z. J., Audretsch, D. B., and Feldman, M. P. (1992). Real effects ofacademic research: comment. American Economic Review, 82(1):363–367.
- Aghion, P., Harris, C., and Vickers, J. (1997). Competition and growth with step-by-step innovation: an example. European Economic Review, 41:771 – 782.
- Aghion, P. and Howitt, P. (1992). A model of growth through creative destruction. Econometrica, 60:323-351.
- Akcigit, U. and Kerr, W. R. (2018). Growth through heterogeneous innovations. Journal of Political Economy, 126(4):1374-1443.
- Allen, T. and Arkolakis, C. (2014). Trade and the topography of the spatial economy. The Quarterly Journal of Economics, 129:1085 - 1140.
- Anselin, L., Varga, A., and Acs, Z. (1997). Local geographic spillovers between university research and high technology innovations. Journal of Urban Economics, 42(3):422-448.
- Audretsch, D. B. and Feldman, M. P. (2004). Knowledge spillovers and the geography of innovation. In V. J., Henderson and J. E., Thisse, editors, Handbook of Regional and Urban Economics., volume 4. Elsevier.
- Baldwin, R., Forslid, R., Martin, P., Ottaviano, G., and Robert-Nicoud, F. (2011). Economic Geography and Public Policy. Princeton University Press.
- Baldwin, R. E. and Forslid, R. (2000). The core-periphery model and endogenous growth: stabilizing and destabilizing integration. *Economica*, 67(267):307–324.
- Bell, A., Chetty, R., Jaravel, X., Petrova, N., and Van Reenen, J. (2019). Who becomes an inventor in america? the importance of exposure to innovation. The Quarterly Journal of Economics, 134: 647 - 713.

- Bloom, N., Schankerman, M., and Van Reenen, J. (2013). Identifying technology spillovers and product market rivarly. *Econometrica*, 81:1347–1393.
- Bond-Smith, S. (2019). The unintended consequences of increasing returns to scale in geographical economics. Technical report, Bankwest Curtin Economics Centre (BCEC), Forthcoming on *Journal of Economic Geography*.
- Caragliu, A. and Nijkamp, P. (2016). Space and knowledge spillovers in european regions: the impact of different forms of proximity on spatial knowledge diffusion. *Journal of Economic Geography*, 16(3):749–774.
- Carlino, G. and Kerr, W. R. (2014). *Agglomeration and Innovation. Working Paper 20367*, National Bureau of Economic Research.
- Combes, P.-P. and Duranton, G. (2006). Labour pooling, labour poaching, and spatial clustering. *Regional Science and Urban Economics*, 36:1–28.
- Cortinovis, N. and van Oort, F. (2019). Between spilling over and boiling down: network-mediated spillovers, local knowledge base and productivity in european regions. *Journal of Economic Geography*, 19(6):1233–1260.
- Dasgupta, P. and Stiglitz, J. (1980). Industrial structure and the nature of innovative activity. *The Economic Journal*, 90(358):266 293.
- De Loecker, J. and Eeckhout, J. (2018). *Global Market Power. Technical Report*, National Bureau of Economic Research.
- Desmet, K., Nagy, D. K., and Rossi-Hansberg, E. (2018). The geography of development. *The Journal of Political Economy*, 126:903 983.
- Eckel, C. and Neary, J. P. (2010). Multi-product firms and flexible manufacturing in the global economy. *The Review of Economic Studies*, 77(1):188–217.
- Etro, F. (2009). Endogenous Market Structure and the Macroeconomy. Springer.
- Feldman, M. P. and Kogler, D. F. (2010). Stylized facts in the geography of innovation. In *Handbook of the Economics of Innovation, Volume* 1, pages 381–410. Elsevier.
- Fujita, M., Krugman, P. R., and Venables, A. J., editors (1999). *The Spatial Economy. Cities, Regions and International Trade.* MIT Press.
- Gertler, M. S. (2003). Tacit knowledge and the economic geography of context, or the undefinable tacitness of being (there). *Journal of economic geography*, 3(1):75–99.
- Glaeser, E. G. and Kohlhase, J. E. (2004). Cities, regions and the decline of transport costs. *Papers in Regional Science*, 83(1):197–228.
- Glaeser, E. L. and Gottlieb, J. D. (2009). The wealth of cities: agglomeration economies and spatial equilibrium in the united states. *Journal of Economic Literature*, 47(4):983–1028.
- Grossman, G. M. and Helpman, E. (1991). *Innovation and Growth in the Global Economy*. MIT press.
- Head, K. and Mayer, T. (2004). The Empirics of Agglomeration and Trade. In J., Henderson and J.F.Thisse, editors, *Handbook of Regional and Urban Economics.*, volume IV, pages 2609–2669. North-Holland.
- Head, K. and Mayer, T. (2010). Gravity, market potential and economic development. *Journal of Economic Geography*, 11(2):281–294.
- Impullitti, G. and Licandro, O. (2018). Trade, firm selection and innovation: the competition channel. *The Economic Journal*, 128(608):189–229.
- Jaffe, A. B. (1989). Real effects of academic research. *The American Economic Review*, 79(5): 957–970.
- Jaffe, A. B., Trajtenberg, M., and Henderson, R. (1993). Geographic localization of knowledge spill-overs as evidenced by patent citations. *The Quarterly Journal of Economics*, 108(3):577–598.
- Kantor, S. and Whalley, A. (2014). Knowledge spillovers from research universities: evidence from endowment value shocks. *Review of Economics and Statistics*, 96(1):171–188.
- Klette, T. J. and Kortum, S. (2004). Innovating firms and aggregate innovation. *Journal of political economy*, 112(5):986–1018.
- Krugman, P. (1980). Scale economies, product differentiation, and the pattern of trade. *The American Economic Review*, 70(5):950 959.
- Krugman, P. (1991). Increasing returns and economic geography. *Journal of Political Economy*, 99(3):483–499.
- Laincz, C. A. and Peretto, P. F. (2006). Scale effects in endogenous growth theory: an error of aggregation not specification. *Journal of Economic Growth*, 11(3):263–288.

- Lychagin, S., Pinkse, J., Slade, M. E., and Reenen, J. V. (2016). Spillovers in space: does geography matter? *The Journal of Industrial Economics*, 64(2):295–335.
- Martin, P. (1999). Public policies, regional inequalities and growth. *Journal of public economics*, 73(1):85–105.
- Martin, P. and Ottaviano, G. I. P. (1999). Growing locations: industry location in a model of endogenous growth. *European Economic Review*, 43(2):281–302.
- Martin, P. and Ottaviano, G. I. P. (2001). Growth and Agglomeration. *International Economic Review*, 42:947–968.
- Miguélez, E. and Moreno, R. (2015). Knowledge flows and the absorptive capacity of regions. *Research Policy*, 44(4):833–848.
- Minerva, G. and Ottaviano, G. I. P. (2009). Endogenous growth theory: agglomeration benefits and transportation costs. In P., Nijkamp and R., Capello, editors, *Handbook of Regional Growth and Development Theories*. Edward Elgar.
- Murata, Y. (2003). Product diversity, taste heterogeneity, and geographic distribution of economic activities:: market vs. non-market interactions. *Journal of Urban Economics*, 53(1):126–144.
- Neary, J. P. (2016). International trade in general oligopolistic equilibrium. *Review of International Economics*, 24(4):669–698.
- Ottaviano, G. I. and Robert-Nicoud, F. (2005). The 'genome' of neg models with vertical linkages: a positive and normative synthesis. *Journal of Economic Geography*, 6(2):113–139.
- Peretto, P. F. (1996). Sunk costs, market structure, and growth. *International Economic Review*, 37(4):895–923.
- Peretto, P. F. (1998). Technological change and population growth. *Journal of Economic Growth*, 3(4):283–311.
- Peretto, P. F. (2003). Endogenous market structure, and the growth and welfare effects of economic integration. *Journal of International Economics*, 60(1):177—201.
- Peretto, P. F. and Connolly, M. (2007). The manhattan metaphor. *Journal of Economic Growth*, 12(4):329–350.
- Redding, S. J. (2011). Economic geography: a review of the theoretical and empirical literature. In D., Bernhofen, R., Falvey, D., Greenaway, and U., Kreickemeier, editors, *The Palgrave Handbook of International Trade*. Palgrave Macmillan.
- Rossi-Hansberg, E., Sarte, P.-D., and Trachter, N. (2018). *Diverging Trends in National and Local Concentration. Technical Report*, National Bureau of Economic Research.
- Smulders, S. and Van de Klundert, T. (1995). Imperfect competition, concentration and growth with firm-specific r & d. *European Economic Review*, 39(1):139–160.
- Spence, M. (1984). Cost reduction, competition, and industry performance. *Econometrica*, 52(1): 101–122.
- Tabuchi, T., Thisse, J. F., and Zhu, X. (2018). Does technological progress magnify regional disparities? *International Economic Review*, 59:647 663.
- Thompson, P. and Fox-Kean, M. (2005). Patent citations and the geography of knowledge spillovers: a reassessment. *American Economic Review*, 95(1):450–460.